Evaluating climate models' cloud feedbacks against expert judgement

Mark D. Zelinka^{1,1}, Stephen A. Klein^{1,1}, and Yi Qin^{2,2}

¹Lawrence Livermore National Laboratory (DOE) ²Lawrence Livermore National Laboratory

November 30, 2022

Abstract

The persistent and growing spread in effective climate sensitivity (ECS) across global climate models necessitates rigorous evaluation of their cloud feedbacks. Here we evaluate several cloud feedback components simulated in 19 climate models against benchmark values determined via an expert synthesis of observational, theoretical, and high-resolution modeling studies. We find that models with smallest feedback errors relative to these benchmark values have moderate total cloud feedbacks (0.4–0.6 Wm $^{-}_{s}^{-2}$ %K $^{-1}$ %) and generally moderate ECS (3–4 K). Those with largest errors generally have total cloud feedback and ECS values that are too large or too small. Models tend to achieve large positive total cloud feedbacks by having several cloud feedback components that are systematically biased high rather than by having a single anomalously large component, and vice versa. In general, better simulation of mean-state cloud properties leads to stronger but not necessarily better cloud feedbacks. The Python code base provided herein could be applied to developmental versions of models to assess cloud feedbacks and cloud errors and place them in the context of other models and of expert judgement in real-time during model development.

Evaluating climate models' cloud feedbacks against expert judgement

Mark D. Zelinka¹, Stephen A. Klein¹, Yi Qin¹, Timothy A. Myers¹

¹Lawrence Livermore National Laboratory

Key Points:

1

2

3

4

5

6

7

8

9

10

• N	Indels with	smallest f	feedback	errors h	nave mod	lerate f	total	cloud	feedbacks	and	ECS
-----	-------------	------------	----------	----------	----------	----------	-------	-------	-----------	-----	-----

• Models with large positive total cloud feedbacks have several systematically highbiased feedback components

• Better simulation of mean-state cloud properties leads to stronger but not necessarily better cloud feedbacks

Corresponding author: Mark D. Zelinka, zelinka1@llnl.gov

11 Abstract

The persistent and growing spread in effective climate sensitivity (ECS) across global 12 climate models necessitates rigorous evaluation of their cloud feedbacks. Here we eval-13 uate several cloud feedback components simulated in 19 climate models against bench-14 mark values determined via an expert synthesis of observational, theoretical, and high-15 resolution modeling studies. We find that models with smallest feedback errors relative 16 to these benchmark values generally have moderate total cloud feedbacks $(0.4-0.6 \text{ Wm}^{-2}\text{K}^{-1})$ 17 and ECS (3-4 K). Those with largest errors generally have total cloud feedback and ECS 18 19 values that are too large or too small. Models tend to achieve large positive total cloud feedbacks by having several cloud feedback components that are systematically biased 20 high rather than by having a single anomalously large component, and vice versa. In gen-21 eral, better simulation of mean-state cloud properties leads to stronger but not neces-22 sarily better cloud feedbacks. The Python code base provided herein could be applied 23 to developmental versions of models to assess cloud feedbacks and cloud errors and place 24 them in the context of other models and of expert judgement in real-time during model 25 development. 26

27 Plain Language Summary

Climate models strongly disagree with each other regarding how much warming 28 will occur in response to increased greenhouse gases in the atmosphere. This is mainly 29 because they disagree on the response of clouds to warming — a process known as the 30 cloud feedback that can amplify or dampen warming initially caused by carbon dioxide. 31 In this study we compare many models' cloud feedbacks to those that have been deter-32 mined by a recent expert assessment of the literature. We find that the models whose 33 cloud feedbacks most strongly disagree with expert assessment tend to have more ex-34 treme cloud feedbacks and hence warm too much or too little in response to carbon diox-35 ide. The models with total cloud feedbacks that are too large do not have a single mas-36 sive feedback component but rather several components that are larger than in other mod-37 els. Models that simulate current-climate clouds that look more like those in nature also 38 simulate stronger amplifying cloud feedbacks, but doing a better job at simulating current-39 climate clouds does not, in general, guarantee a better simulation of cloud feedbacks. 40

41 **1** Introduction

Cloud feedback — the change in cloud-induced top-of-atmosphere radiation anoma-42 lies with global warming — is the primary driver of differences in effective climate sen-43 sitivity (ECS) across global climate models (GCMs). This has been the case for all ex-44 isting model intercomparisons, starting with Cess et al. (1989, 1990) and continuing to 45 the most recent collection of models as part of CMIP6, the 6th phase of the Coupled Model 46 Intercomparison Project (M. D. Zelinka et al., 2020; Eyring et al., 2016). Despite sub-47 stantial progress in understanding, diagnosing, modeling, and observationally constrain-48 ing cloud feedbacks from a variety of approaches, the spread in cloud feedbacks across 49 GCMs has remained substantial through the decades and actually increased in CMIP6 50 relative to CMIP5 (M. D. Zelinka et al., 2020). Moreover, strengthened cloud feedback 51 particularly for extratropical low clouds — is the primary reason for the increase in 52 average climate sensitivity in CMIP6 relative to CMIP5, as well as for the emergence of 53 models with very high ECS above the upper limit of the *likely* range (1.5-4.5 K) reported 54 in the fifth assessment report of the Intergovernmental Panel on Climate Change (M. D. Zelinka 55 et al., 2020; Flynn & Mauritsen, 2020; M. Collins et al., 2013). 56

Given the need for models to reliably predict future climate and the fact that cloud feedbacks strongly affect their ability to do so makes it imperative to evaluate models' cloud feedbacks against some form of ground truth. Such an evaluation is now possible because quantitative values of individual cloud feedbacks (and their uncertainties) were
recently determined based on an expert synthesis of theoretical, observational, and highresolution cloud modeling evidence. This synthesis was conducted as part of a broader
assessment of climate sensitivity, in which three semi-independent lines of evidence (process studies, historical climate record, and paleoclimate record) were brought together
in a Bayesian framework to place robust bounds on Earth's climate sensitivity (Sherwood
et al., 2020).

Our goals in this work are several-fold. First, we evaluate GCM cloud feedback components against those assessed in Sherwood et al. (2020). This allows us to answer several questions, including: Do models with extremely large or small climate sensitivities have cloud feedback components that are erroneous? If so, which component(s)? How are cloud feedbacks in CMIP6 — and their biases with respect to expert assessment changing from CMIP5? Are some models getting the "right" total cloud feedback via erroneous components that compensate?

Second, we investigate whether the fidelity with which models simulate present-74 day cloud properties is linked to their cloud feedbacks and to the fidelity with which their 75 cloud feedbacks agree with expert judgement. A key question is whether better simu-76 lation of present-day cloud properties leads to cloud feedbacks that are better aligned 77 with expert judgement. This is particularly relevant because aspects of the cloud sim-78 ulation in many high-ECS CMIP6 models are in many cases considered superior to those 79 in CMIP5 (Gettelman et al., 2019; Bodas-Salcedo et al., 2019), yet holistic aspects of the 80 climate simulation in these models appear inferior to their lower-ECS counterparts (Zhu 81 et al., 2020, 2021; Tokarska et al., 2020; Nijsse et al., 2020) 82

Finally, we provide a code base to compute cloud feedbacks and error metrics for all of the assessed categories, and visualize them in a multi-model context. This will allow, for example, model developers to evaluate cloud feedbacks in developmental versions of their models against expert judgement, other models, and other variants of their model, providing them with detailed information about a key process affecting their model's climate sensitivity.

⁸⁹ 2 Data and Methods

We are primarily interested in cloud feedbacks in response to CO_2 -induced global 90 warming, so we make use of abrupt CO_2 quadrupling experiments conducted with fully-91 coupled GCMs in CMIP5 and CMIP6 (abrupt-4xC02). We first compute cloud radia-92 tive anomalies at the top-of-atmosphere (TOA) by multiplying cloud fraction anoma-93 lies with cloud radiative kernels (M. D. Zelinka et al., 2012a, 2012b). The cloud fraction 94 anomalies needed for this calculation are reported in a matrix of 7 cloud top pressure 95 (CTP) categories by 7 visible optical depth (τ) categories matching the categorization 96 of the International Satellite Cloud Climatology Project (ISCCP; Rossow & Schiffer, 1999). 97 These matrices are produced by the ISCCP simulator (Klein & Jakob, 1999; M. Webb 98 et al., 2001), referred to as clisccp in CMIP parlance. Cloud radiative kernels quan-99 tify the sensitivity of top-of-atmosphere radiative fluxes to small cloud fraction pertur-100 bations in each of these 49 cloud types. Hence the product of the two yields the radi-101 ation anomaly from each cloud type, which can be summed over the entire matrix to pro-102 vide the total cloud radiative anomalies at a given location. Because of the reliance on 103 cliscop, we are limited in this study to those models (listed in Table 1) that have suc-104 cessfully implemented the Cloud Feedback Model Intercomparison Project (CFMIP) Ob-105 servation Simulator Package (COSP; Bodas-Salcedo et al., 2011). As will be evident be-106 low, these models exhibit cloud feedbacks spanning nearly the full range of values pro-107 duced in the full ensemble of CMIP5 and CMIP6 models analyzed in M. D. Zelinka et 108 al. (2020), and we therefore consider this subset to be a sufficiently representative sam-109 ple of model diversity. 110

Symbol	Model	Reference	Years
a	CCSM4	Gent et al. (2011)	1-104
b	CanESM2	Arora et al. (2011)	1-21 / 121-140
с	HadGEM2-ES	W. J. Collins et al. (2011)	1-20 / 122-140
d	MIROC-ESM	S. Watanabe et al. (2011)	1-20 / 121-140
е	MIROC5	M. Watanabe and others (2010)	1-20 / 121-140
f	MPI-ESM-LR	Stevens et al. (2013)	1-20 / 121-140
g	MRI-CGCM3	Yukimoto et al. (2012)	1-20 / 121-140
Н	CNRM-CM6-1	Voldoire et al. (2019)	1-150
Ι	CNRM-ESM2-1	Séférian et al. (2019)	1-150
J	CanESM5	Swart et al. (2019)	1-150
Κ	E3SM-1-0	Golaz et al. (2019)	1-150
L	GFDL-CM4	Held et al. (2019)	1-150
Μ	HadGEM3-GC31-LL	K. D. Williams et al. (2018)	1-150
Ν	IPSL-CM6A-LR	Boucher et al. (2020)	1-150
Ο	IPSL-CM6A-LR-INCA	Boucher et al. (2020)	1-150
Р	MIROC-ES2L	Hajima et al. (2020)	1-150
Q	MIROC6	Tatebe et al. (2019)	1-150
R	MRI-ESM2-0	Yukimoto et al. (2019)	1-150
S	UKESM1-0-LL	Sellar et al. (2019)	1-150

Table 1. Models used in this study. CMIP5 and CMIP6 models are indicated with lowercase and upper-case symbols, respectively. Years within the abrupt-4xC02 simulation with data available to analyze are indicated.

Anomalies are computed with respect to the contemporaneous pre-industrial control (piControl) simulation, with three exceptions: CNRM-CM6-1, CNRM-ESM2-1, and IPSL-CM6A-LR-INCA did not archive cliscop from the piControl simulation, so we take this field from piClim-control, a 30-year long atmosphere-only simulation that uses sea-surface temperatures (SSTs) and sea ice concentrations fixed at the model-specific piControl climatology (Pincus et al., 2016).

We compute cloud feedbacks by regressing annual mean cloud-radiative anomalies on annual and global mean surface air temperature anomalies over the duration of the 150-year abrupt-4xC02 experiment containing all necessary data. In CMIP6, clisccp output is available throughout the full duration of the run, whereas in CMIP5 it is typically only available for two non-contiguous 20-year periods, one at the beginning and one at the end of the run (Table 1).

M. D. Zelinka et al. (2012a) validated cloud feedbacks computed using the cloud 123 radiative kernel (CRK) methodology against independent estimates derived as the ad-124 justed change in cloud radiative effect (ΔCRE_{adi} ; Shell et al., 2008; Soden et al., 2008) 125 for six CMIP3 models. Here we update this comparison using the CMIP5 and CMIP6 126 models analyzed in this study. We compare CRK-derived cloud feedbacks with the ΔCRE_{adj} 127 and approximate partial radiative perturbation (APRP; Taylor et al., 2007)-derived val-128 ues computed in M. D. Zelinka et al. (2020). Six ΔCRE_{adj} feedbacks are provided based 129 on the adjustments from the non-cloud radiative kernels of Soden et al. (2008), Shell et 130 al. (2008), Block and Mauritsen (2013), Huang et al. (2017), Pendergrass et al. (2018), 131 and Smith et al. (2018). APRP provides only the SW component, but it additionally pro-132 vides estimates of SW cloud amount, scattering, and absorption feedbacks, allowing us 133 to compare to the CRK-derived SW amount and optical depth components. Figure S1 134 shows the multi-model mean zonal mean SW and LW cloud feedbacks from these three 135 techniques, along with their across-model correlations, and Figure S2 scatters the global 136 mean CRK-derived and non-CRK-derived feedback values against each other. The CRK-137 derived feedbacks are in excellent agreement with the ΔCRE_{adj} and APRP feedbacks, 138 for both the spatial characteristics of the multi-model mean and the across-model cor-139 relation of the zonal and global means. This confirms the validity of the CRK technique 140 for estimating cloud feedback. 141

We focus in this study on feedbacks estimated from abrupt-4xC02 experiments so 142 as to stay consistent with Sherwood et al. (2020), but have repeated all calculations using Atmospheric Model Intercomparison Project (amip) experiments with imposed +4K144 SST perturbations that are spatially uniform (amip-p4K) and patterned (amip-future4K), 145 as described in the CFMIP protocol (M. J. Webb et al., 2017). Feedbacks in these sim-146 ulations were computed as cloud radiation anomalies normalized by global mean surface 147 air temperature anomalies between the +4K experiments and the control amip exper-148 iment. All basic conclusions reported in this study are insensitive to whether we con-149 sider feedbacks diagnosed in amip-p4K, amip-future4K, or abrupt-4xCO2 experiments. 150

To distinguish feedbacks occurring in regions of large-scale ascent from those oc-151 curring in regions of large-scale descent over tropical oceans, we aggregate (with area-152 weighting) all monthly control and perturbed climate fields over the tropical oceans into 153 10-hPa wide bins of 500 hPa vertical pressure velocity (ω_{500}) following Bony et al. (2004). 154 Anomalies between perturbed and control climates are then performed in ω_{500} space rather 155 than geographic space when computing tropical marine ascent/descent feedbacks. The 156 resulting feedbacks can be further broken down into dynamic, thermodynamic, and co-157 variance terms (see Bony et al., 2004), but for the purposes of this study, we will con-158 sider only their sum, and will further aggregate these to "ascent regions" where $\omega_{500} <$ 159 0 and "descent regions" where $\omega_{500} \ge 0$. 160

Following M. D. Zelinka et al. (2016), we separately quantify feedbacks arising from low, boundary layer clouds and from non-low, free tropospheric clouds, hereafter referred

to as "low" and "high" cloud feedbacks, respectively. This is done by performing the cloud 163 feedback calculations using only restricted parts of the clisccp histogram: CTPs > 680164 hPa for low clouds and CTPs ≤ 680 hPa for high clouds. Within these subsets, the cloud 165 feedback is further broken down into (1) the "amount" component due to change in to-166 tal cloud fraction holding CTP and τ distribution fixed; (2) the "altitude" component 167 due to the change in CTP distribution holding total fraction and τ distribution fixed; 168 and (3) the "optical depth" component due to the change in τ distribution holding the 169 total fraction and CTP distribution fixed (M. D. Zelinka et al., 2013, 2016). 170

Passive satellite-based measurements – like those mimicked by the ISCCP simulator used in this study – provide unobscured cloud fractions visible from space. This means that low-clouds may be hidden and revealed by changes in high-cloud cover. This complicates interpretation of low-cloud feedbacks, since high-cloud changes are aliased to an unknown extent into low-cloud feedbacks. To avoid this potential source of misinterpretation, we express the standard low-level cloud feedbacks as a sum of three terms following Scott et al. (2020) and Myers et al. (2021):

$$low = low_{unobsc} + \Delta obsc + cov$$

low_{unobsc} is the "true" low-cloud feedback occurring in regions that are not obscured by 179 upper-level clouds and are unaffected by changes in obscuration, which we further break 180 down into amount, altitude, optical depth, and residual components. Δ obsc is the "obscuration-181 induced" component of low-cloud feedback arising entirely from changes in upper-level 182 cloud fraction that reveal or hide low-level clouds. It is therefore by definition solely an 183 "amount" component, so we absorb it into the high-cloud amount feedback. The covari-184 ance term, cov, is typically very small. To summarize, the total cloud feedback can be 185 expressed as: 186

$$total = \sum_{i} high_i + \sum_{i} low_{unobsc,i} + cov,$$

where $i \in \{\text{amount, altitude, optical depth, residual}\}$ components, and the high cloud amount component includes the $\Delta \text{obsc component.}$

In Table 2, we list the central value and 1- σ uncertainty of the cloud feedback com-190 ponents assessed in Sherwood et al. (2020) and describe how we compute them in GCMs. 191 We also provide a matrix in Figure S3 to help visualize the feedback components that 192 are computed in this study. A large amount of observational evidence, based mainly on 193 inter-annual variability, was used to provide quantitative values for the assessed total cloud 194 feedback and several of its individual components. In addition, process-resolving mod-195 els in the form of large eddy simulations were a key piece of evidence for the strength 196 of tropical marine low cloud feedback, while guidance from theoretical understanding un-197 derlies the assessed high cloud altitude, tropical anvil, and land-cloud amount feedbacks. 198 Many of the expert assessed cloud feedbacks are independent of any GCM results, but 199 the assessed central value and uncertainty for the high cloud altitude, land cloud amount, 200 and middle latitude marine low cloud amount feedbacks were derived at least partially 201 from GCMs, albeit a collection that included pre-CMIP5 models that are excluded here 202 and that excluded some recently-published CMIP6 models that are included here. Com-203 paring GCM results to expert-assessed values can therefore be thought of as a quick and 204 economical way of evaluating model feedbacks against the very wide body of evidence 205 that forms the basis of the expert-assessed cloud feedbacks. 206

Values of effective climate sensitivity (ECS) are taken from M. D. Zelinka et al. (2020), updated to include recently-available models. These ECS values are computed in a manner consistent with the cloud feedbacks, by regressing global and annual mean TOA net radiative flux anomalies on global and annual mean surface air temperature anomalies over the 150-year duration of the abrupt-4xC02 experiment. Anomalies are computed with respect to the contemporaneous piControl simulation, except in IPSL-CM6A-LR-INCA, for which we use piClim-control because no piControl fields are available.

Expert-Assessed Feedbacks		Calculation in GCMs			
Name	Value	Components	Surface	Regime	Region
1. high cloud altitude	0.2 ± 0.10	high-cloud altitude	all	all	N06-S06
2. tropical marine low-cloud	0.25 ± 0.16	sum of low-cloud amount, altitude, $\&$ optical depth	ocean	descent	30S-30N
3. tropical anvil cloud area	-0.2 ± 0.20	sum of high-cloud amount $\&$ optical depth	ocean	ascent	30S-30N
4. land cloud amount	0.08 ± 0.08	sum of high- and low-cloud amount	land	all	N06-S06
5. middle-latitude marine low-cloud amount	0.12 ± 0.12	low-cloud amount	ocean	all	30-60N/S
6. high-latitude low-cloud optical depth	0.00 ± 0.10	low-cloud optical depth	all	all	40-70N/S
7. sum of assessed	0.45 ± 0.33	sum of items 1–6			
8. total cloud feedback	0.45 ± 0.33	total cloud feedback	all	all	N06-S06
9. implied unassessed	N/A	item 8 minus item 7			

terest with weighting by the fractional area of the globe represented. As explained the text, high-cloud amount feedbacks include the Δ obsc term and all low-cloud component is computed in GCMs in this study. Feedbacks are computed at each spatial location (or ω_{500} bin as appropriate), then summed over the region of in-Central value and $1-\sigma$ uncertainty of the cloud feedback components assessed in Sherwood et al. (2020) (in Wm⁻²K⁻¹), and description of how each feedbacks are computed using low_{unobsc} components. Table 2.

Finally, for each model we compute a radiatively-relevant cloud property error met-214 ric, E_{NET} , using Equation 5 of Klein et al. (2013). First, cloud fraction errors are com-215 puted by differencing climatological ISCCP simulator cloud fraction histograms from amip 216 simulations and the ISCCP HGG observational climatology (Young et al., 2018). Both 217 modeled and observed climatologies are computed over the 26-year period January 1983 218 to December 2008, when all model simulations and observations overlap, but error met-219 rics are very insensitive to the time period considered. Second, these errors are multi-220 plied by net (LW+SW) cloud radiative kernels, thereby weighting them by their corre-221 sponding net TOA radiative impact. Third, this product is aggregated into six cloud types: 222 optically intermediate and thick clouds at low, middle, and high levels. These are then 223 squared, averaged over the six categories, summed (with area weighting) over month, lon-224 gitude, and latitude between 60° S and 60° N, and the square root is taken. Finally, this 225 scalar value is normalized by the accumulated space-time standard deviation of observed 226 radiatively-relevant cloud properties, defined analogously. This process yields a single 227 scalar error metric, E_{NET} , in each model that quantifies the spatio-temporal error in cli-228 matological cloud properties for clouds with $\tau > 3.6$, weighted by their net TOA radia-229 tive impact. We acknowledge that evaluation against ISCCP observations is a limited 230 viewpoint on the quality of models' cloud simulations — one that may change if using 231 other cloud datasets, like those derived from active sensors. 232

- 233 3 Results
- 234

3.1 GCM Cloud Feedbacks Evaluated Against Expert-Assessed Values

In Figure 1, cloud feedbacks from 7 CMIP5 and 12 CMIP6 models are compared 235 with the assessed values for feedback categories listed in Table 2. Each feedback value 236 is scaled by the fractional area of the globe occupied by that cloud type such that sum-237 ming all components yields the global mean feedback. Each marker is color-coded by its 238 ECS, with the color boundaries corresponding to the 5th, 17th, 83rd, and 95th percentiles 239 of the Baseline posterior PDF of ECS from Table 10 of Sherwood et al. (2020). In Ta-240 ble 3, we list the GCM values and highlight any values that lie outside of the very likely 241 (90%) and *likely* (66%) confidence intervals of expert judgement with double and sin-242 gle asterisks, respectively. Supplementary Figures 4-22 are identical to Figure 1, but with 243 individual models highlighted in each figure for better discrimination. 244

All but seven models fall within the *likely* range assessed for the high cloud alti-245 tude feedback and the multi-model means are very close to the central assessed value. 246 However, some models have weak high cloud altitude feedbacks that lie below the lower 247 bound of the likely (MRI-CGCM3 and MIROC6) and very likely (MIROC5 and MIROC-248 ES2L) confidence intervals, and some have strong high cloud altitude feedbacks that lie 249 above the upper bound of the likely (HadGEM2-ES and CanESM5) and very likely (E3SM-250 1-0) confidence intervals. This feedback component has the greatest number of models 251 (3) lying outside of the assessed very likely range; these are the same three models that 252 lie outside the assessed very likely range for total cloud feedback. Such wide inter-model 253 variation is noteworthy for a feedback having a strong theoretical basis and both obser-254 vational and high-resolution modeling support. 255

²⁵⁶ Consistent with Klein et al. (2017), the distribution of modeled tropical marine low
²⁵⁷ cloud feedback values favors the low end of the expert assessed value. Only one model
²⁵⁸ (CanESM5) exceeds the central expert assessed value, and several models' values lie be²⁵⁹ low the lower bound of the *likely* (MIROC5, MRI-CGCM3, HadGEM3-GC31-LL, MIROC²⁶⁰ ES2L, and MIROC6) and very likely (CCSM4) confidence intervals.

In contrast, all models underestimate the strength of the negative anvil cloud feed back, relative to the central value assessed in Sherwood et al. (2020). Eight models (MRI CGCM3, CNRM-CM6-1, CNRM-ESM2-1, E3SM-1-0, HadGEM3-GC31-LL, IPSL-CM6A-

Table 3.	Individual cloud feedback components (in $Wm^{-2}K^{-1}$), cloud feedback RMSE values (in $Wm^{-2}K^{-1}$), net radiatively-relevant cloud property error met-
rics (E_{NET}	r; unitless), and effective climate sensitivities (ECS; K) for all models analyzed in this study. Expert-assessed central values and uncertainties of cloud
feedback c	components are also provided. Any model values that lie outside of the very likely (90%) and likely (66%) confidence intervals of expert judgement are
denoted w	vith double and single asterisks, respectively.

0		4	\$										
Model	Variant	High	Marine	Tropical	Land	Midlat	Hilat	Unassessed	Sum	Total	RMSE	E_{NET}	ECS
		Alt.	Low	Anvil	Amt.	Low Amt.	Low Tau		Assessed				
a) CCSM4	r2i1p1	0.11	-0.05**	-0.07	0.08	0.07	-0.03	-0.01	0.12^{*}	0.11^{*}	0.14	1.42	2.94
b) CanESM2	rlilpl	0.27	0.14	-0.06	0.06	0.05	-0.05	0.07	0.42	0.49	0.09	1.09	3.70
c) HadGEM2-ES	r1i1p1	0.30^{*}	0.15	-0.02	0.07	0.11	-0.06	0.03	0.56	0.59	0.10	0.89	4.60^{*}
d) MIROC-ESM	r1i1p1	0.17	0.15	-0.04	0.10	0.13	-0.14*	0.09	0.38	0.47	0.10	N/A	4.65^{*}
e) MIROC5	rlilpl	0.00^{**}	0.07^{*}	-0.14	0.03	-0.04*	-0.05	-0.13	-0.11**	-0.24**	0.13	1.59	2.71
f) MPI-ESM-LR	r1i1p1	0.17	0.22	-0.05	0.08	0.11	-0.16*	0.03	0.37	0.40	0.09	1.21	3.63
g) MRI-CGCM3	rlilpl	0.10^{*}	0.07^{*}	0.03^{*}	0.03	0.04	-0.06	-0.02	0.22	0.20	0.13	0.93	2.61
CMIP5 Average		0.16	0.11	-0.05	0.07	0.07	-0.08	0.01	0.28	0.29	0.11	1.19	3.55
CMIP5 1- σ		0.10	0.08	0.05	0.02	0.05	0.05	0.07	0.20	0.27	0.02	0.25	0.78
H) CNRM-CM6-1	r1i1p1f2	0.27	0.06^{*}	0.02^{*}	0.04	0.05	-0.01	0.12	0.42	0.54	0.13	0.70	4.90^{**}
I) CNRM-ESM2-1	r1i1p1f2	0.23	0.04^{*}	0.02^{*}	0.03	0.02	-0.01	0.11	0.34	0.45	0.13	0.71	4.79^{**}
J) CanESM5	r1i1p2f1	0.30^{*}	0.27	-0.06	0.05	0.09	-0.03	0.17	0.62	0.78^{*}	0.08	0.91	5.62^{**}
K) E3SM-1-0	rlilplfl	0.38^{**}	0.21	0.01^{*}	0.09	0.21	-0.02	0.24	0.88^{*}	1.12^{**}	0.12	0.80	5.31^{**}
L) GFDL-CM4	rlilplfl	0.19	0.17	-0.12	0.09	0.19	-0.05	0.11	0.46	0.57	0.06	0.80	3.89
M) HadGEM3-GC31-LL	r1i1p1f3	0.20	0.09^{*}	0.03^{*}	0.07	0.25^{*}	-0.01	0.12	0.64	0.76	0.12	0.79	5.55^{**}
N) IPSL-CM6A-LR	rlilplfl	0.29	0.13	0.02^{*}	0.13	0.21	-0.04	0.05	0.76	0.81^{*}	0.12	1.08	4.70^{*}
O) IPSL-CM6A-LR-INCA	rlilplfl	0.27	0.13	0.02^{*}	0.14	0.21	-0.04	0.05	0.73	0.78^{*}	0.12	N/A	4.13^{*}
P) MIROC-ES2L	r1i1p1f2	0.01^{**}	0.06^{*}	-0.19	0.05	-0.01*	-0.03	-0.11	-0.12**	-0.23**	0.12	1.57	2.66
Q) MIROC6	rlilplfl	*60.0	0.05^{*}	-0.08	0.10	-0.05*	-0.04	-0.11	0.06^{*}	-0.05*	0.13	1.44	2.60
R) MRI-ESM2-0	r1i1p1f1	0.24	0.15	-0.06	0.01	0.12	-0.04	0.03	0.43	0.45	0.08	0.96	3.13
S) UKESM1-0-LL	r1i1p1f2	0.23	0.10	0.02^{*}	0.06	0.25^{*}	-0.02	0.15	0.65	0.80^{*}	0.12	0.81	5.36^{**}
CMIP6 Average		0.23	0.12	-0.03	0.07	0.13	-0.03	0.08	0.49	0.56	0.11	0.96	4.39
CMIP6 1- σ		0.09	0.07	0.07	0.04	0.10	0.01	0.10	0.28	0.36	0.02	0.28	1.05
CMIP5/6 Average		0.20	0.12	-0.04	0.07	0.11	-0.05	0.05	0.41	0.46	0.11	1.04	4.08
CMIP5/6 1- σ		0.10	0.07	0.06	0.03	0.09	0.04	0.10	0.27	0.36	0.02	0.29	1.04
WCRP Central WCRP 1- σ		$0.2 \\ 0.10$	0.25 0.16	-0.2 0.20	0.08	$0.12 \\ 0.12$	$0.0 \\ 0.10$	N/A N/A	0.45 0.33	0.45 0.33			



Figure 1. Cloud feedback components estimated from climate model simulations and as assessed in Sherwood et al. (2020). For each component, the individual model values are indicated with symbols, the multi-model means are indicated with green (CMIP5) and purple (CMIP6) bars, and the expert assessed *likely* and *very likely* confidence intervals are indicated with black errorbars. Model symbols are color-coded by ECS with color boundaries corresponding to the edges of the *likely* and *very likely* ranges of the Baseline posterior PDF of ECS from Sherwood et al. (2020). Identical figures highlighting each individual model are provided in Figures S4-S22.

LR, IPSL-CM6A-LR-INCA, and UKESM1-0-LL) have positive anvil feedbacks that place them above the upper bound of the assessed *likely* confidence interval.

All models lie within the assessed *likely* range for the land cloud amount feedback, while all but five models (MIROC5, HadGEM3-GC31-LL, MIROC-ES2L, MIROC6, and UKESM1-0-LL) lie within the assessed *likely* range of the middle latitude marine low cloud amount feedback.

Whereas the central estimate of the high latitude low cloud optical depth feedback 270 from the assessment is 0, all models simulate a negative feedback. All but two models 271 (MIROC-ESM and MPI-ESM-LR) fall within the *likely* assessed range, however. In the 272 multi-model average, the negative feedback values are more than halved in CMIP6 rel-273 ative to CMIP5, bringing CMIP6 models into better agreement with expert judgement. 274 This may be related to a weakened cloud phase feedback owing to improved simulation 275 of mean-state cloud phase (Bodas-Salcedo et al., 2019; Gettelman et al., 2019; M. D. Zelinka 276 et al., 2020; Flynn & Mauritsen, 2020). The inter-model spread in this feedback com-277 ponent has also dramatically decreased in CMIP6. 278

The unassessed feedback is near zero on average across all models, consistent with it being assigned a value of zero in the expert assessment. However, its across-model standard deviation and its CMIP5-to-CMIP6 increase in multi-model average are larger than all other individual components except the high cloud altitude feedback. Contributors to this feedback will be discussed in greater detail in Section 3.5.

The sum of all six assessed feedback components is positive in all but two models 284 (MIROC5 and MIROC-ES2L) and exhibits substantially more inter-model spread than 285 any individual component comprising it. Its standard deviation ($\sigma = 0.27 \text{ Wm}^{-2}\text{K}^{-1}$) 286 is also larger than would exist if the feedback components comprising it were uncorre-287 lated across models (σ if summing individual uncertainties in quadrature = 0.20 Wm⁻²K⁻¹), 288 as discussed further in Section 3.2. While the multi-model mean value is close to the expert-289 assessed value, some models lie below the lower bound of the assessed *likely* (CCSM4 and 290 MIROC6) and very likely (MIROC5 and MIROC-ES2L) confidence intervals, and E3SM-291 1-0 lies above the upper bound of the assessed *likely* confidence interval. 292

The total cloud feedback, which is the sum of assessed and unassessed components, 293 has a larger standard deviation than would occur if these two components were uncor-294 related. Owing to this correlation, all but four models (MIROC-ESM, MPI-ESM-LR, 295 CNRM-ESM2-1, and MRI-ESM2-0) exhibit degraded agreement with expert assessment 296 once accounting for their unassessed feedbacks. In addition to the models that fell out-297 side the *likely* and *very likely* ranges for the sum of assessed feedbacks, there are now 298 four new models (CanESM5, IPSL-CM6A-LR, IPSL-CM6A-LR-INCA, and UKESM1-299 0-LL) that lie above the upper bound of the assessed *likely* confidence interval, and E3SM-300 1-0 has now moved above the upper bound of the assessed very likely confidence inter-301 val. 302

Unsurprisingly, models with larger total cloud feedback tend to have higher ECS. 303 All five models with total cloud feedbacks above the upper limit of the expert-assessed 304 likely range (CanESM5, E3SM-1-0, IPSL-CM6A-LR, IPSL-CM6A-LR-INCA, and UKESM1-305 0-LL) are part of CMIP6. These models also have ECS values above 3.9 K, the upper 306 limit of the expert-assessed *likely* ECS range, and all but IPSL-CM6A-LR and IPSL-CM6A-307 LR-INCA have ECS values above 4.7 K, the upper limit of the very likely ECS range. 308 However, two models with ECS > 3.9 K (HadGEM2-ES, MIROC-ESM) and even three with ECS > 4.7 K (CNRM-CM6-1, CNRM-ESM2-1, and HadGEM3-GC31-LL) have to-310 tal cloud feedbacks within the *likely* range, indicating that other non-cloud feedbacks are 311 pushing these models to very high ECS. No models considered here — even those whose 312 cloud feedbacks lie below the lower limit of the *likely* and very *likely* total cloud feed-313 back confidence bound — have ECS values below 2.6 K, the lower limit of the Sherwood 314

et al. (2020) assessed *likely* range. In general, too-large cloud feedbacks seem to guarantee too-large ECS, but too-small cloud feedbacks do not guarantee too-small ECS. Also, too-large ECS can arise even without too-large cloud feedbacks.

Turning now to the multi-model mean cloud feedback components, we see that the mean total cloud feedback is roughly twice as large in CMIP6 than in CMIP5, qualitatively consistent with M. D. Zelinka et al. (2020), who assessed a much larger collection of models. This occurs because the high cloud altitude, midlatitude marine low cloud amount, high latitude low cloud optical depth, and unassessed feedbacks all become more positive, on average, in CMIP6. The other feedbacks change very little on average.

All multi-model mean assessed feedback components lie within the respective expert-324 assessed *likely* range. They also lie very close to the central assessed values, with two ex-325 ceptions: The tropical marine low cloud feedback averaged across all models (0.12 ± 0.07 326 $Wm^{-2}K^{-1}$) is about half as large as assessed (0.25 ± 0.16 $Wm^{-2}K^{-1}$), and the trop-327 ical anvil cloud area feedback averaged across all models is close to zero (-0.04 ± 0.06 328 $\text{Wm}^{-2}\text{K}^{-1}$), whereas it was assessed to be moderately negative $(-0.20 \pm 0.20 \text{ Wm}^{-2}\text{K}^{-1})$. 329 For these two components, GCM values were not used to inform the expert judgement 330 value, but rather they were based upon observations and, in the case of tropical marine 331 low cloud feedbacks, large eddy simulations that resolve many of the cloud processes that 332 must be parameterized in GCMs (see Table 1 of Sherwood et al., 2020). 333

3.2 Correlations Among GCM Cloud Feedbacks

334

The previous section provided several indications that models with large positive total cloud feedbacks tend to have systematically higher cloud feedbacks for *all* components rather than having a single anomalously strong positive component, and vice versa for models with small or negative total cloud feedbacks. We quantify this more rigorously in this section by diagnosing the correlation structure among the individual components.

All individual cloud feedback components are positively correlated with the total 341 cloud feedback, especially the high cloud altitude, midlatitude marine low cloud amount. 342 and unassessed feedbacks (Figure 2a, column 1). While the tropical marine low cloud 343 feedback is significantly correlated with the total, it is markedly weaker than for several 344 other components, which is surprising given previous findings that low latitude marine 345 low clouds in regions of moderate subsidence drive inter-model spread in climate sen-346 sitivity (Bony & Dufresne, 2005). The discrepancy may arise from the relatively small 347 subset of models considered here, but it also may be related to the precise definition of 348 low-cloud types: Taking the sum of stratocumulus and trade cumulus cloud feedbacks 349 diagnosed in Myers et al. (2021) using different meteorological criteria than employed 350 here as an alternative estimate of tropical marine low-cloud feedback, we find a larger 351 correlation (r=0.80) with total cloud feedback. 352

The positive correlations between individual components and the total cloud feed-353 back is expected: If all the models were distributed randomly for each feedback compo-354 nent, one would expect the models with largest total cloud feedback to be the ones that 355 most consistently lie on the positive tail of all components. To demonstrate this, we gen-356 erated normal distributions with 10,000 samples matching the multi-model mean and 357 standard deviation for each of the six assessed and one unassessed components and re-358 peated the above calculations on these random data. All individual components are sig-359 nificantly positively correlated with their sum, with correlation strengths proportional 360 361 to the individual component variances (Figure 2b, column 1).

The prevalence of strong and significant positive correlations among individual feedback components seen in the actual model data is, however, not expected from chance. This leads to (1) individual components being more strongly correlated with the total



Figure 2. Matrix showing the across-model correlation among all cloud feedback components for (a) actual model data and (b) synthetic normally-distributed data with means and standard deviations equal to those of the models for each feedback component. Correlations that are significantly different from zero at the 95% confidence level are indicated with an asterisk.

cloud feedback and (2) a wider spread in the total cloud feedback than would occur if 365 individual components were uncorrelated. Models with large positive total cloud feed-366 backs tend to have systematically larger-than-average cloud feedbacks across multiple 367 components rather than being generally near-average but having a single large compo-368 nent. E3SM-1-0, for example, has the largest positive total cloud feedback, and its feed-369 back values are among the largest values in all categories except the land cloud feedback 370 (Figure S14 and Table 3). Conversely, models like MIROC5 with negative total cloud 371 feedbacks tend to have cloud feedbacks on the left tail of the distribution for all com-372 ponents (Figure S8 and Table 3). Consistent with this, we find that most models with 373 near-average total cloud feedbacks have components that are systematically near-average 374 rather than having several components with extreme values of opposing sign that counter 375 each other. One exception is CNRM-ESM2-1, which has feedbacks on the high tail of 376 the model distribution for some components and on the low tail for others (Figure S12 377 and Table 3). 378

That all of the *significant* correlations in Figure 2a are positive might suggest that 379 they are linked by a physical mechanism rather than arising from tuning artifacts. As 380 will be shown in Section 3.5, high-cloud feedbacks are among the largest components of 381 the unassessed feedback; hence it is plausible that the positive correlations among the 382 unassessed, high-cloud altitude, and anvil feedbacks reflect a shared physical mechanism 383 involving high clouds. Other large positive correlations (e.g., between high-cloud alti-384 tude and tropical and middle latitude marine low-cloud amount) are harder to rational-385 ize. We discuss further implications of all of these correlations in Section 3.4. 386

387

3.3 Metrics of Overall Cloud Feedback Errors

To assess the overall skill of each model in matching the expert-assessed cloud feedback components, we compute a single cloud feedback error metric for each model as the root mean square error (RMSE) with respect to the central expert judgement value over all six assessed feedback components of Sherwood et al. (2020). Each model's cloud feedback RMSE is provided in Table 3 and is plotted against total cloud feedback in Figure 3.

³⁹⁴ CMIP5 and CMIP6 models exhibit both high and low cloud feedback RMSE val-³⁹⁵ ues, and the multi-model mean RMSE values are the same for both ensembles (Table



Figure 3. Total cloud feedback scattered against cloud feedback RMSE, with expert *likely* and *very likely* ranges of total cloud feedback indicated with horizontal shading. Models are denoted by the symbols listed in Table 3 and are colored according to their (a) ECS values and (b) net radiatively-relevant cloud property error metric, E_{NET} .

3). Although the three best-performing models in this measure are CMIP6 models, there 396 is no systematic tendency for CMIP6 models to be performing better than CMIP5 mod-397 els with respect to expert judgement. For models from the same modelling centers that 398 can be tracked between the two generations, the same number of models show degraded 399 performance as improved performance in this measure: MIROC-ES2L [P] and the two 400 UKMO models (HadGEM3-GC31-LL [M] and UKESM1-0-LL [S]) have higher RMSE 401 than their predecessors (MIROC-ESM [d], and HadGEM2-ES [c]), whereas CanESM5 402 [J], MIROC6 [Q], and MRI-ESM2-0 [R] have lower RMSE than their predecessors (CanESM2 403 [b], MIROC5 [e], and MRI-CGCM3 [g]). 404

The seven models with smaller-than-average cloud feedback errors (i.e., RMSE \leq 405 $0.11 \text{ Wm}^{-2}\text{K}^{-1}$ have moderate (0.4–0.6 Wm⁻²K⁻¹) total cloud feedbacks, except for 406 CanESM5 [J], which has a total cloud feedback of 0.8 $Wm^{-2}K^{-1}$. All but three of these 407 models have moderate (3–4 K) ECS values, the exceptions being HadGEM2-ES [c], MIROC-408 ESM [d], and CanESM5 [J], which have ECS values above 4.5 K. This makes sense given 409 that the expert-assessed value of total cloud feedback, which has the greatest leverage 410 on ECS, led to moderate values of ECS in Sherwood et al. (2020). Of the seven mod-411 els with below-average feedback errors, GFDL-CM4 [L], MRI-ESM2-0 [R], and CanESM2 412 b] are the only ones for which all assessed feedbacks lie within the expert *likely* range 413 (Figures S15, S21, and S5, respectively; Table 3). Put simply, they get the right answer 414 for the right reasons. 415

Models with too-large or too-small total cloud feedbacks and ECS tend to have larger-416 than-average cloud feedback RMSE values. That is, the models that lie farthest from 417 the horizontal dashed line tend to be located on the right side of Figure 3. All five mod-418 els with small total cloud feedback (< 0.2 $\rm Wm^{-2}K^{-1}$) and small ECS (< 3 K) have cloud 419 feedback components that are systematically biased low relative to expert judgement, 420 giving them larger-than-average RMSE. Most models with large total cloud feedback and 421 large ECS have cloud feedback components that are systematically biased high relative 422 to expert judgement, also giving them larger-than-average RMSE. Of the nine models 423 with ECS > 4.5 K, only HadGEM2-ES [c], MIROC-ESM [d], and CanESM5 [J] have below-424 average RMSE value. CCSM4 [a] has the highest RMSE of all models considered despite 425 lying within the assessed *likely* range for five components (Figure S4; Table 3). 426

Two models (CNRM-CM6-1 [H] and CNRM-ESM2-1 [I]) have total cloud feedbacks very close to the central value of the expert assessment but larger-than-average RMSE values. They achieve reasonable total cloud feedbacks partly through having low-biased



Figure 4. (a) Total cloud feedback and (b) cloud feedback RMSE scattered against net radiatively-relevant cloud property error metric, E_{NET} . Models are denoted by the symbols listed in Table 3 and are colored green for CMIP5 and purple for CMIP6. Expert *likely* and *very likely* ranges of total cloud feedback indicated with horizontal shading in (a). Correlations that are significant at 95% confidence are indicated with an asterisk.

tropical marine low cloud feedbacks that counteract their high-biased tropical anvil cloud
area feedbacks (Figures S11-12; Table 3). Put simply, they get the right answer for the
wrong reasons.

GFDL-CM4, CanESM5, MRI-ESM2-0, and CanESM2 remain the four models with
 lowest RMSE regardless of whether we use feedbacks derived from abrupt-4xC02 or amip-p4K
 experiments.

436 437

3.4 Relationship Between Cloud Feedbacks and Mean-State Cloud Property Errors

The fidelity with which models simulate mean-state radiatively-relevant cloud prop-438 erties is strongly and significantly correlated with total cloud feedback (Figure 4a). We 439 show this result for the net radiatively-relevant cloud property error (E_{NET}) , but it is 440 also strong and significant for the SW-radiation error as well as the cloud property er-441 ror without radiative weighting (not shown). This result is consistent with Figure 11 of 442 Klein et al. (2013), but now the relationship holds across two ensembles of models (CMIP5 443 and CMIP6). Given that E_{NET} is an aggregated metric, we also tested whether the an-444 ticorrelation persists when considering relationships between individual cloud feedbacks 445 and cloud-type specific E_{NET} values (e.g., between midlatitude marine low-cloud amount 446 feedback and mean-state errors for midlatitude marine low-clouds). This anticorrelation 447 continues to hold for all but the land cloud amount feedback, albeit with weaker corre-448 lation coefficients (not shown). While caution is necessary given the relatively small sam-449 ple size, an important question is why better simulating present-day cloud properties is 450 associated with larger cloud feedbacks. We leave this as an open question for future re-451 search. 452

On average, mean-state cloud properties are simulated better in CMIP6 than in 453 CMIP5 (Figure 4a; Table 3). Six CMIP6 models now have smaller error values than the 454 smallest exhibited in CMIP5. For models from the same modeling center than can be 455 tracked, all but one has improved in this measure from CMIP5 to CMIP6. Specifically, 456 marked improvement is seen from CanESM2 [b] to CanESM5 [J], from HadGEM2-ES 457 [c] to HadGEM3-GC31-LL [M] and UKESM1-0-LL [S], and from MIROC5 [e] to MIROC6 458 [Q], whereas MRI-ESM2-0 [R] has very slightly degraded mean-state clouds relative to 459 MRI-CGCM3 [g]. 460

It is often implicitly assumed by model developers and model analysts that the de-461 gree to which a model's clouds resembles reality can be used as a basis to trust their re-462 sponse to climate change. In Figure 4b, we test this assumption by comparing the agree-463 ment with expert judgment for cloud feedbacks (encapsulated in RMSE) to the agreement with observations of the present-day climatological distribution of clouds and their 465 properties (encapsulated in E_{NET}). While the correlation between these two metrics is 466 positive, it is very weak and not significant at 95% confidence. Moreover, many mod-467 els with small mean-state cloud errors have cloud feedback errors that are as large or larger 468 than models with large mean-state errors, indicating that improved simulation of mean-469 state cloud properties does not necessarily lead to improved cloud feedbacks with respect 470 to expert judgment. The weak correlation also holds for relationships between RMSE 471 and components of E_{NET} corresponding to individual cloud feedbacks (not shown). 472

In Figure 3b, models are color-coded by E_{NET} , allowing for a simultaneous assess-473 ment of how well models simulate mean-state cloud properties and match expert judg-474 ment of total cloud feedback and its components. From this it is evident that most of 475 the models with small mean-state errors (yellow shading) have large cloud feedback er-476 rors and several lie above the upper limit of the *likely* range of total cloud feedback (i.e., 477 in the top-right portion of the diagram). The one exception is GFDL-CM4 [L], which 478 achieves low cloud feedback RMSE, low values of E_{NET} , and total cloud feedback near 479 the central value of expert judgement. 480

While realistic mean-state cloud properties may not guarantee that a model sim-481 ulates more reliable cloud feedbacks, the models with worst mean-state cloud proper-482 ties (i.e., $E_{\rm NET} > 1.3$) all have poor agreement with the expert-assessed total cloud feed-483 back and/or its components (see models at top right of Figure 4b). This is also evidenced 484 by the fact that most of the models with large mean-state errors (purple/black shading) 485 have large cloud feedback RMSE and lie below the lower limit of the *likely* range of to-486 tal cloud feedback (i.e., in the bottom-right part of Figure 3b). This suggests that sim-487 ulating poor mean-state cloud properties precludes a model from simulating cloud feed-488 backs in agreement with expert judgement. In other words, better simulation of mean-489 state cloud properties may be a necessary but insufficient criterion for simulating more 490 trustworthy cloud feedbacks. 491

This finding has support in recent literature. Mülmenstädt et al. (2021) showed 492 that a model with better mean-state cloud properties could have greater biases in its cli-493 mate responses owing to compensating errors in cloud and precipitation processes. As 494 noted in that study, fidelity in simulating mean-state clouds alone is an insufficient con-495 straint on a model's feedback because of the many different combinations of process rep-496 resentations that can lead to equally valid representations of mean-state clouds. Since 497 these process representations can all differ in their sensitivity to warming, the cloud feed-498 back is not uniquely determined by mean-state properties, and improving the represen-499 tation of the mean-state (especially at the expense of the process-level) does not guar-500 antee that feedbacks will be more reliably simulated. This notion is supported by the 501 fact that the set of model parameters driving the variance in mean-state extratropical 502 cloud radiative effect across members of the HadGEM3-GA7.05 perturbed physics en-503 semble differ from those driving the variance in its cloud feedback (Tsushima et al., 2020). 504 A corollary to this are the many examples in which models with better "bottom-up" pro-505 cess representation more poorly satisfy "top-down" constraints like the observed histor-506 ical global mean temperature evolution (Golaz et al., 2013; Suzuki et al., 2013), expert-507 assessed magnitude of aerosol indirect effects (Jing & Suzuki, 2018) or paleoclimate states 508 (Zhu et al., 2020, 2021) 509

3.5 GCM Cloud Feedbacks in Unassessed Categories

510

Sherwood et al. (2020) only assessed quantitative values for a selection of well-studied 511 cloud feedbacks, so it is important to know whether any of the unassessed feedbacks are 512 substantial. Examining these feedback components is important as it may guide where 513 future research with observations, process-resolving models, and theory is needed to fur-514 ther constrain GCMs' cloud feedbacks. Figure 5 shows a breakdown of explicitly-computed 515 feedbacks that were not assessed in Sherwood et al. (2020). There are an infinite num-516 ber of ways of breaking down these components, but our strategy was to quantify those 517 that complement the assessed feedbacks, either in altitude or geographic space, to the 518 extent possible. For example, we quantify the low cloud altitude feedback since the high 519 cloud feedback is an assessed category, and we quantify the low cloud optical depth feed-520 back between 30 and 90 degrees latitude but excluding the 40–70 degree zone where it 521 was already assessed. The sum of these closely reproduces the implied unassessed feed-522 backs in Figure 1 (not shown). See Figure S3 for a matrix that helps to visualize and 523 rationalize the discretization made. 524

The multi-model mean unassessed cloud feedback transitions from being $0.01 \text{ Wm}^{-2}\text{K}^{-1}$ on average in CMIP5 to $0.08 \text{ Wm}^{-2}\text{K}^{-1}$ on average in CMIP6. The largest shift occurs for the multi-model mean extratropical high cloud optical depth component, which transitions from a negative to a weak positive value. This component, along with the tropical marine ascent low-cloud amount plus optical depth component exhibit the largest inter-model spread among all unassessed categories, and may be worthwhile targets for future expert assessment.

There are a few models whose unassessed feedbacks sum to a value that is large 532 relative to their total and/or combined assessed feedbacks and worth examining in greater 533 detail. MIROC5, MIROC-ES2L, and MIROC6 exhibit strong negative unassessed cloud 534 feedbacks (with values $< -0.10 \text{ Wm}^{-2}\text{K}^{-1}$) that are comparable in magnitude to the 535 sum of their assessed feedbacks. MIROC5 and MIROC6 have strong negative low-cloud 536 amount plus optical depth components in tropical marine ascent regions, while MIROC-537 ES2L has strong negative high-cloud amount and optical depth components in tropical 538 marine subsidence regions. All three of these models have moderately negative extra-539 tropical high-cloud optical depth feedbacks as well. Two CMIP6 models (CanESM5 and 540 E3SM-1-0) have positive unassessed feedbacks that exceed $0.15 \text{ Wm}^{-2}\text{K}^{-1}$ — the multi-541 model mean plus standard deviation. This occurs because of several systematically pos-542 itive components, the largest of which is the 0.11 $\mathrm{Wm}^{-2}\mathrm{K}^{-1}$ extratropical high-cloud 543 optical depth component in E3SM-1-0. 544

⁵⁴⁵ 4 Discussion and Conclusions

We have evaluated cloud feedback components simulated in 19 CMIP5 and CMIP6 546 models against benchmark values determined via an expert synthesis of observational, 547 theoretical, and high-resolution modeling studies (Sherwood et al., 2020). We found that, 548 in general, models that most closely match the expert-assessed values across several cloud 549 feedback components have moderate total cloud feedbacks $(0.4-0.6 \text{ Wm}^{-2}\text{K}^{-1})$ and mod-550 erate ECS (3–4 K). In contrast, models with largest feedback errors with respect to ex-551 pert assessment generally have total cloud feedbacks and climate sensitivities that are 552 too large or too small. 553

There is no evidence that CMIP6 models simulate cloud feedbacks in better agreement with expert judgement than do CMIP5 models. While the three best models in our error metric are CMIP6 models, all models with total cloud feedbacks above the upper limit of the expert-assessed *likely* range are part of CMIP6 and have ECS values above 3.9 K, the upper limit of the expert-assessed *likely* ECS range. However, the converse is not true: several models with high ECS have total cloud feedbacks within the *likely*



Figure 5. As in Figure 1, but for cloud feedback components that were not assessed in Sherwood et al. (2020). Note the x-axis spans a range that is only a third of that in Figure 1.

range. This means that large cloud feedback ensures a high ECS, but high ECS can emerge even with moderate cloud feedbacks, a result consistent with M. J. Webb et al. (2013) for CMIP3 models. More generally, having 2xCO₂ radiative forcing and feedbacks in agreement with expert judgement does not guarantee that a model's ECS will be in agreement with expert judgement because the latter is further constrained by evidence from the paleoclimate and historical records (Sherwood et al., 2020).

On average, and for most individual modeling centers, mean-state cloud properties are better simulated in CMIP6. Better simulation of mean-state cloud properties is strongly and significantly correlated with larger total cloud feedback. The reasons for this remain to be investigated, but it is consistent with emergent constraint studies involving mean-state properties of clouds or their environment, nearly all of which point to higher-than-average cloud feedbacks and climate sensitivities (Volodin, 2008; Trenberth & Fasullo, 2010; Fasullo & Trenberth, 2012; Sherwood et al., 2014; Tian, 2015; Brient et al., 2016; Siler et al., 2018).

But more skillful simulation of mean-state cloud properties does not guarantee more 574 skillful simulation of cloud feedbacks, and many models with small mean-state errors have 575 large cloud feedback errors with respect to expert judgment. In general, better simula-576 tion of mean-state cloud properties leads to stronger but not necessarily better cloud feed-577 backs. GFDL-CM4, which has the smallest cloud feedback error, small mean-state cloud 578 property error, and a total cloud feedback near the expert-assessed central value, is the 579 exception to this rule. Skill at simulating mean-state cloud properties appears to be a 580 necessary but not sufficient criterion for simulating realistic cloud feedbacks. 581

Models with large positive total cloud feedbacks tend to have systematically higher 582 cloud feedbacks for all components rather than having a single anomalously strong pos-583 itive component, and vice versa for models with small or negative total cloud feedbacks. This means, for example, that there is no single feedback that all high ECS models are 585 exaggerating. However, if there is some physical relationship causing the correlation be-586 tween individual feedback components, this may imply that constraining one component 587 would have knock-on effects across several components. In this case, feedbacks from mul-588 tiple cloud types could be constrained with less evidence than would be needed if they 589 were uncorrelated, and changing one aspect of a model might systematically change the 590 feedbacks from multiple cloud types, making it easier to improve its cloud feedbacks. Es-591 tablishing and understanding the physical basis of correlations among feedback compo-592 nents and their potential linkages with mean-state cloud properties is important future 593 work. 594

The high latitude low-cloud optical depth feedback has shifted from being robustly negative across CMIP5 models, with some models simulating moderately strong negative feedbacks below the expert-assessed *likely* range, to a much weaker negative feedback in CMIP6, with the models tightly clustered about it. This represents a shift towards better agreement with expert judgement (also seen in Myers et al., 2021), and may be tied to reductions in super-cooled liquid biases in the latest models (Bodas-Salcedo et al., 2019; Gettelman et al., 2019; M. D. Zelinka et al., 2020).

Results from several individual cloud feedback components raise important questions and motivate future investigation:

604

605

606

- The high cloud altitude feedback strength varies widely across models, despite its firm theoretical basis and support from observational analyses and high-resolution modeling. This motivates further work to pin down causes of inter-model spread and to eliminate sources of bias in this feedback.
- Although we found that the tropical marine low cloud feedback simulated by most models lies at the low end of the expert-assessed *likely* range, recent observational constraints support slightly lower values (Cesana & Del Genio, 2021; Myers et al.,

6112021; Ceppi & Nowack, 2021) owing in part to a better discrimination between612strong stratocumulus feedbacks and weaker trade cumulus feedbacks. If incorpo-613rated into a future assessment, the expert value of this feedback could be revised614downward, likely resulting in a better alignment between it and the multi-model615mean. To the extent that the assessed confidence bounds also narrow, however,616the models with very weak tropical marine low cloud feedbacks may still lie be-617low the expert judgement range.

- Despite the wide uncertainty in its expert-assessed value, eight models have pos-618 itive tropical anyil cloud feedbacks that place them above the upper bound of the 619 assessed *likely* confidence interval. This discrepancy between models and expert 620 judgment can be traced to the disagreement between models and observations in 621 the sensitivity of tropical TOA radiation and deep convective cloud properties to 622 interannual fluctuations in surface temperature found in the studies of Mauritsen 623 and Stevens (2015) and I. N. Williams and Pierrehumbert (2017), which were in-624 fluential in establishing the expert-assessed value. Much uncertainty remains sur-625 rounding the processes controlling tropical anvil cloud fraction and its changes with 626 warming, and the fidelity with which GCMs can simulate them (Bony et al., 2016; 627 Hartmann, 2016; Seeley et al., 2019; Wing et al., 2020; Gasparini et al., 2021). 628
- Cloud feedback components that were not assessed in Sherwood et al. (2020), though 629 summing to zero on average across models, have substantial inter-model spread 630 and partly drive the increase in multi-model average cloud feedback from CMIP5 631 to CMIP6. Of these, the extratropical high cloud optical depth component exhibits 632 the largest increase. This, along with the aforementioned uncertainties surround-633 ing high cloud altitude and anvil cloud feedbacks highlights the need for further 634 observational analyses, process-resolving modeling, and theoretical studies target-635 ing high cloud feedbacks. 636

We have provided Python code that performs all calculations and generates all vi-637 sualizations presented in this study. The code is also easily modified to accommodate 638 comparisons between GCM cloud feedbacks and the similar but not identical breakdown 639 of cloud feedback components that is used in the 6th Assessment report of the IPCC. 640 We envision that this code could be applied to perturbed parameter or perturbed physics 641 ensembles and to developmental versions of models to assess cloud feedbacks and cloud 642 errors and place them in the context of other models and of expert judgement in real-643 time during model development. This may be particularly valuable in less computation-644 ally expensive prescribed SST perturbation experiments that are routinely performed 645 during model development. Despite their simpler design, these "Cess-type" experiments 646 effectively capture the feedbacks present in fully coupled experiments (Ringer et al., 2014). 647 So doing could help modelers to identify and correct erroneous cloud feedbacks that lead 648 to biased climate sensitivity prior to the model being frozen, thereby increasing the re-649 liability of the model for policy-relevant climate projections (e.g., Voosen, 2021). 650

651 Acknowledgments

Python code to perform all calculations and produce all figures and tables in this manuscript 652 is available at https://doi.org/10.5281/zenodo.5206838 (M. Zelinka, 2021a) and is 653 being incorporated into the PCMDI Metrics Package (Doutriaux et al., 2018), available 654 at https://doi.org/10.5281/zenodo.1414560. CMIP5 and CMIP6 ECS values are 655 available at https://doi.org/10.5281/zenodo.5206851 (M. Zelinka, 2021b). ISCCP 656 HGG cloud data is provided by NOAA/NCEI at https://www.ncei.noaa.gov/thredds/ 657 catalog/cdr/isccp_hgg_agg/files/catalog.html. We acknowledge the World Climate 658 Research Programme, which, through its Working Group on Coupled Modelling, coor-659 dinated and promoted CMIP. We thank the climate modeling groups for producing and 660 making available their model output, the Earth System Grid Federation (ESGF) for archiv-661 ing the data and providing access, and the multiple funding agencies who support CMIP 662

and ESGF. This work was supported by the U.S. Department of Energy (DOE) Regional

- and Global Modeling Analysis program area and was performed under the auspices of
- the DOE by Lawrence Livermore National Laboratory under Contract DE-AC52-07NA27344.
- We are grateful for stimulating discussions with Leo Donner, Chris Golaz, Yoko Tsushima,
- and Mark Webb, and for the helpful comments from three anonymous reviewers.

668 References

669	Arora, V. K., Scinocca, J. F., Boer, G. J., Christian, J. R., Denman, K. L., Flato,
670	G. M., Merryfield, W. J. (2011). Carbon emission limits required to satisfy
671	future representative concentration pathways of greenhouse gases. Geophys.
672	Res. Lett., 38. doi: 10.1029/2010GL046270
673	Block, K., & Mauritsen, T. (2013). Forcing and feedback in the MPI-ESM-LR cou-
674	pled model under abruptly quadrupled CO2. Journal of Advances in Modeling
675	Earth Systems, 5(4), 676–691. doi: 10.1002/jame.20041
676	Bodas-Salcedo, A., Mulcahy, J. P., Andrews, T., Williams, K. D., Ringer, M. A.,
677	Field, P. R., & Elsaesser, G. S. (2019). Strong Dependence of Atmospheric
678	Feedbacks on Mixed-Phase Microphysics and Aerosol-Cloud Interactions in
679	HadGEM3. Journal of Advances in Modeling Earth Systems, 11(6), 1735–
680	1758. doi: 10.1029/2019ms001688
681	Bodas-Salcedo, A., Webb, M. J., Bony, S., Chepfer, H., Dufresne, J. L., Klein, S. A.,
682	John, V. O. (2011). COSP Satellite simulation software for model assess-
683	ment. Bulletin of the American Meteorological Society, 92(8), 1023–1043. doi:
684	10.1175/2011bams 2856.1
685	Bony, S., & Dufresne, J. L. (2005). Marine boundary layer clouds at the heart of
686	tropical cloud feedback uncertainties in climate models. Geophys. Res. Lett.,
687	32. doi: 10.1029/2005GL023851
688	Bony, S., Dufresne, J. L., Treut, H. L., Morcrette, J. J., & Senior, C. (2004). On
689	dynamic and thermodynamic components of cloud changes. Climate Dyn., 22,
690	71–68. doi: 10.1007/s00382-003-0369-6
691	Bony, S., Stevens, B., Coppin, D., Becker, T., Reed, K. A., Voigt, A., & Medeiros,
692	B. (2016). Thermodynamic control of anvil cloud amount. Proceedings of the
693	National Academy of Sciences. doi: 10.1073/pnas.1601472113
694	Boucher, O., Servonnat, J., Albright, A. L., Aumont, O., Balkanski, Y., Bastrikov,
695	V., Vuichard, N. (2020). Presentation and Evaluation of the IPSL-
696	CM6A-LR Climate Model. Journal of Advances in Modeling Earth Sys-
697	<i>tems</i> , 12(7), e2019MS002010. Retrieved 2021-07-17, from https://agupubs
698	.onlinelibrary.wiley.com/doi/abs/10.1029/2019MS002010 (_eprint:
699	https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2019MS002010) doi:
700	10.1029/2019MS002010
701	Brient, F., Schneider, T., Tan, Z., Bony, S., Qu, X., & Hall, A. (2016, July).
702	Shallowness of tropical low clouds as a predictor of climate models' re-
703	sponse to warming. <i>Climate Dynamics</i> , 47(1), 433–449. Retrieved
704	2021-08-04, from https://doi.org/10.1007/s00382-015-2846-0 doi:
705	10.1007/s00382-015-2846-0
706	Ceppi, P., & Nowack, P. (2021, July). Observational evidence that cloud feedback
707	amplifies global warming. Proceedings of the National Academy of Sciences,
708	118(30). Retrieved 2021-08-09, from https://www.pnas.org/content/118/
709	30/e2026290118 (Publisher: National Academy of Sciences Section: Physical
710	Sciences) doi: 10.10/3/pnas.2026290118
711	Cesana, G. V., & Del Genio, A. D. (2021, March). Observational constraint on cloud
712	teedbacks suggests moderate climate sensitivity. Nature Climate Change,
713	11(3), 213-218. Retrieved 2021-03-09, from https://www.nature.com/
714	articles/s41558-020-009/0-y doi: 10.1038/s41558-020-009/0-y
715	Cess, R. D., Potter, G. L., Blanchet, J. P., Boer, G. J., Genio, A. D. D.,

 pretation of climate feedback processes in 19 atmospheric general circulation models. Journal of Gcophysical Research: Atmospheres, 95(1010), 16601–16615. Retrieved 2021-08-10, from https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/JD0951D10p16601 cess, R. D., Potter, G. L., Blanchet, J. P., Boer, G. J., Ghan, S. J., Kiehl, J. T., Yagai, I. (1989). Interpretation of Cloud-Climate Feedback as Produced by 14 Atmospheric General Circulation Models. Science, 245(4917), 513–516. doi: 10.126/science.245.4917.513 Collins, M., Knutti, R., Arblaster, J., Dufresne, JL., Fichefet, T., Friedlingstein, P., Wehner, M. (2013). Long-term Climate Change: Projections, Commitments and Irreversibility. In Climate Change: 10:31: The Physical Science Basis. Contribution of Working Group 1 to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change: Cambridge United Kingdom and New York, NY, USA:: Cambridge University Press. Collins, W. J., Bellouin, N., Doutriaux-Boucher, M., Gedney, N., Hinton, P. H. T., Hughes, J., Woodward, S. (2011). Development and evaluation of an Earth-system model - HadGEM2. Geosci. Model Dev. Discuss., 4, 997–1062. Doutriaux, C., Gleckler, P., Durack, P. J., Lee, J., Covey, C., Sperber, K., jservonnat (2018, September). PCMDI/pendi.metrics: PMP Version 1:2. Zenodo. Retrieved 2021-08-16, from https://zenodo.org/record/1414560 Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMP6) experimental design and organization. Geosci. Model Dev., 9(5), 1937–1958. doi: 10.5194/gmd-9-1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Subtropical Subisdence in Climate Scinsitivity. Science, 338(6108), 792–794. doi: 10.1126/science.1220/7829/2020/ doi: https://doi.org/10.5194/acp-20-7829-2020 Gasparini, B., Rasch, P. J., Har	716	Déqué, M., Zhang, MH. (1990). Intercomparison and inter-
 culation models. Journal of Geophysical Research: Atmospheres, 95 (D10), 16601–16615. Retrieved 2021-08-10, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/JD095iD10p16601 cb: 10.1029/JD095iD10p16601 Cess, R. D., Potter, G. L., Blanchet, J. P., Boer, G. J., Ghan, S. J., Kiehl, J. T., Yagai, I. (1989). Interpretation of Cloud-Climate Feedback as Produced by 14 Atmospheric General Circulation Models. Science, 245 (4917), 513–516. doi: 10.1129/science.245.4917.513 Collins, M., Knutti, R., Arblaster, J., Dufresne, JL., Fichefet, T., Friedlingstein, P., Wehner, M. (2013). Long-term Climate Change: Projections, Commitments and Irreversibility. In <i>Climate Change</i>. 2013: The Physical Science Basis. Contribution of Working Group 1 to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, United Kingdom and New York, NY, USA.: Cambridge University Press. Collins, W. J., Belouin, N., Doutriaux-Boucher, M., Gedney, N., Hinton, P. H. T., Hughes, J., Woodward, S. (2011). Development and evaluation of an Earth-system model - HadGEM2. Geosci. Model Dev. Discuss. 4, 997–1062. Doutriaux, C., Gleckler, P., Durack, P. J., Lee, J., Covey, C., Sperber, K., jservonnat (2018, September). <i>PCMDI/pcmdi.metrics: PMP Version 1.2</i>. Zenodo. Retrieved 2021-08-16, from https://zenodo.org/record/1414560 Eyring, V., Boay, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taykor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. Geosci. Model Dev., 9(5), 1937-1958. doi: 10.5194/gmd-91937-2016 Faynin, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and historical warming evolution in recent coupled model ensembles. Atmospheric Chemistry and Physics, 20(13), T829–7842. Retrieved 2021-03-10, from https://acgupus.com/doi/di/10.1029/2020D033487 Gent, P., R., Danabasoglu	717	pretation of climate feedback processes in 19 atmospheric general cir-
 95(D10), 16601-16615. Retrieved 2021-08-10, from https://agyupbs.onlinelibrary.wiley.com/doi/abs/10.1029/JD095iD10p16601 doi: 10.1029/JD095iD10p16601 doi: 10.1029/JD095iD10p16601 Cess, R. D., Potter, G. L., Blanchet, J. P., Boer, G. J., Ghan, S. J., Kichl, J. T., Yagai, I. (1989). Interpretation of Cloud-Climate Feedback as Produced by 14 Atmospheric General Circulation Models. Science, 245(4917), 513-516. doi: 10.1126/science.245.4917.513 Collins, M., Knutti, R., Arblaster, J., Dufresne, JL., Fichefet, T., Friedlingstein, P., Wehner, M. (2013). Long-term Climate Change: Projections, Committuents and Irreversibility. In Climate Change 2013: The Physical Science Basis. Contribution of Working Group 1 to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press. Collins, W. J., Bellouin, N., Doutriaux-Boucher, M., Gedney, N., Hinton, P. H. T., Hughes, J., Woodward, S. (2011). Development and evaluation of an Earth-system model - HadGEM2. Geosci. Model Dev. Discuss., 4, 997-1062. Doutriaux, C., Gleckler, P., Durack, P. J., Lee, J., Covey, C., Sperber, K., jservounat (2018, September). PCMDI/pendi.metrics: PMP Version 1.2. Zenodo. Retrieved 2021-08-16, from https://zenodo.org/record/1414560 Eyring, V., Bony, S., Mechl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. Geosci. Model Des., 9(5), 1937-1958. doi: 10.5194/gmdJ-91372016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Subtropical Subsidence in Climate Sensitivity. Science, 338(6108), 792-794. doi: 10.1126/science.1220/7829/2020/ doi: https://doi.org/10.5194/acb/20/7829/2020/ Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C.	718	culation models. Journal of Geophysical Research: Atmospheres,
 onlinelibrary.viley.com/doi/abs/10.1029/JD095iD10p16601 (.eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/JD095iD10p16601 doi: 10.1029/JD095iD10p16601 Cess, R. D., Potter, G. L., Blanchet, J. P., Boer, G. J., Ghan, S. J., Kiehl, J. T., Yagai, I. (1989). Interpretation of Cloud-Climate Feedback as Produced by 14 Atmospheric General Circulation Models. Science, 245(4917), 513–516. doi: 10.1126/science.245.4917.513 Collins, M., Kmuti, R., Arblaster, J., Dufresne, JL., Fichefet, T., Friedlingstein, P., Wehner, M. (2013). Long-term Climate Change: Projections, Com- mitments and Irreversibility. In Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Interpowermental Panel on Climate Change. Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press. Collins, W. J., Bellouin, N., Doutriaux-Boucher, M., Gedney, N., Hinton, P. H. T., Hughes, J., Woodward, S. (2011). Development and evaluation of an Earth-system model HadGEM2. Genesic. Model Dev. Discuss, 4, 997–1062. Doutriaux, C., Gleckler, P., Durack, P. J., Lee, J., Covey, C., Sperber, K., jser- vonnat (2018, September). PCMDI/pemdi.metrics: PMP Version 1.2. Zenodo. Retrieved 2021-08-16, from https://zenodo.org/record/1414560 doi: 10.5281/zenodo.1414560 Eyring, V., Bony, S., Mechl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. Geosci. Model Dev., 9(5), 1937–1958. doi: 10.5194/gmd-9-1937-2016 Fasulto, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Sub- tropical Subidence in Climate Sensitivity. Science, 338(6108), 792–794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and histori- cal warning bengistic, 20(13), 7829	719	95(D10), 16601–16615. Retrieved 2021-08-10, from https://agupubs
 https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/JD095iD10p16601 cess, R. D., Potter, G. L., Blanchet, J. P., Boer, G. J., Ghan, S. J., Kiehl, J. T., Yagai, I. (1989). Interpretation of Cloud-Climate Feedback as Produced by 14 Atmospheric General Circulation Models. Science, 245 (4917), 513–516. doi: 10.1126/science.245.0417.513 Collins, M., Knutti, R., Arblaster, J., Dufresne, JL., Fichefet, T., Friedlingstein, P.,, Wehner, M. (2013). Long-term Climate Change 2013: The Physical Science Basis. Contribution of Working Group 1 to the Fifth Assessment Report of the Intergovermmental Panel on Climate Change. Cambridge, United Kingdom and New York, NY, USA.: Cambridge University Press. Collins, W. J., Bellouin, N., Doutriaux-Boucher, M., Gedney, N., Hinton, P. H. T., Hughes, J., Woodward, S. (2011). Development and evaluation of an Earth-system model - HadGEM2. Geosci. Model Dev. Discuss., 4, 997–1062. Doutriaux, C., Gleckler, P., Durack, P. J., Lee, J., Covey, C., Sperber, K., jservonnat (2018, September). PCMDI/pendi.metrics: PMP Version 1.2. Zenodo. Retrieved 2021-08-16, from https://zenodo.org/record/1414560 doi: 10.5281/zenodo.1414560 Eyring, V., Bony, S., Mechl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. Geosci. Model Dev., 9(5), 1937–1958. doi: 10.5194/gmd-9.1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Subtropical Subsidence in Climate Sensitivity. Science, 338 (6108), 792-794. doi: 10.1126/science.1227165 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and historical warming evolution in recent coupled model ensembles. Atmospheric, Chemistry and Physics, 20(13), 7829-7842. Retrieved 2021-03-10, from https://agupuse.oninelibrary.wiley.com/doi/abs/10.1029/	720	.onlinelibrary.wiley.com/doi/abs/10.1029/JD095iD10p16601 (_eprint:
 doi: 10.1029/JD095iD10p1601 Cess, R. D., Potter, G. L., Blanchet, J. P., Boer, G. J., Ghan, S. J., Kichl, J. T., Yagai, I. (1989). Interpretation of Cloud-Climate Feedback as Produced by 14 Atmospheric General Circulation Models. Science, 245(4917), 513–516. doi: 10.1126/science.245.4917.513 Collins, M., Knutti, R., Arblaster, J., Dufresne, JL., Fichefet, T., Friedlingstein, P., Wehner, M. (2013). Long-term Climate Change: Projections, Com- mitments and Inreversibility. In Climate Change: 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press. Collins, W. J., Bellouin, N., Doutriaux-Boucher, M., Gedney, N., Hinton, P., H. T., Hughes, J., Woodward, S. (2011). Development and evaluation of an Earth-system model - HadGEM2. Genesic. Model Dev. Discuss., 4, 997–1062. Doutriaux, C., Gleckler, P., Durack, P. J., Lee, J., Covey, C., Sperber, K., jser- vonnat (2018, September). PCMDI/pcmdi.metrics: PMP Version 1.2. Zenodo. Retrieved 2021-08-16, from https://zenodo.org/record/1414560 doi: 10.5281/zenodo.1414560 Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. Geosci. Model Dcr., 9(5), 1937–1958. doi: 10.5194/gmd-9-1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Sub- tropical Subidence in Climate Sensitivity. Science, 338(6108), 792–794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and histori- cal warning evolution in recent coupled model ensembles. Atmospheres, Ichen- sity and Physics, 20(13), 7829–7842. Retrieved 2021-03-10, from https://agupus .copernicus.org/articles/20/7829/2020	721	https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/JD095iD10p16601)
 Cess, R. D., Potter, G. L., Blanchet, J. P., Boer, G. J., Ghan, S. J., Kiehl, J. T., Yagai, I. (1989). Interpretation of Cloud-Climate Feedback as Produced by 14 Atmospheric General Circulation Models. Science, 245 (4917), 513–516. doi: 10.1126/science.245.4917.513 Collins, M., Knutti, R., Arblaster, J., Dufresne, JL., Fichefet, T., Friedlingstein, P., Wehner, M. (2013). Long-term Climate Change: Projections, Com- mitments and Irreversibility. In Climate Change. 2013: The Physical Science Basis. Contribution of Working Group 1 to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press. Collins, W. J., Bellouin, N., Doutriaux-Boucher, M., Gedney, N., Hinton, P. H. T., Hughes, J., Woodward, S. (2011). Development and evaluation of an Earth-system model - HadGEM2. Geosci. Model Den. Discuss., 4, 997–1062. Doutriaux, C., Gleckler, P., Durack, P. J., Lee, J., Covey, C., Sperber, K., jser- vonnat (2018, September). PCMDI/pcmdi.metrics: PMP Version 1.2. Zenodo. Retrieved 2021-08-16, from https://zenodo.org/record/1414560 doi: 10.5281/zenodo.1414560 Eyring, V., Bony, S., Mechl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMFP6) experimental design and organization. Geosci. Model Dex., 9(5), 1937–1958. doi: 10.5194/gmd-9-1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Sub- tropical Subidence in Climate Sensitivity. Science, 338(6108), 792–794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and histori- cal warming evolution in recent coupled model emsembles. Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://acp .copernicus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/ acp=20-7829-2020 Gas	722	doi: $10.1029/JD095iD10p16601$
 Yagai, I. (1980). Interpretation of Cloud-Climate Feedback as Produced by 14 Atmospheric General Circulation Models. Science, 245(4917), 513–516. doi: 10.1126/science.245.4917.513 Collins, M., Knutti, R., Arblaster, J., Dufresne, JL., Fichefet, T., Friedlingstein, P., Wehner, M. (2013). Long-term Climate Change: Projections, Com- mitments and Irreversibility. In Climate Change 2013: The Physical Science Basis. Contribution of Working Group 1 to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, United Kingdom and New York, NY, USA.: Cambridge University Press. Collins, W. J., Bellouin, N., Doutriaux-Boucher, M., Gedney, N., Hinton, P. H. T., Hughes, J., Woodward, S. (2011). Development and evaluation of an Earth-system model - HadGEM2. Geosci. Model Dev. Discuss, 4, 997–1062. Doutriaux, C., Gleckler, P., Durack, P. J., Lee, J., Covey, C., Sperber, K., jser- vonnat (2018, September). PCMDI/pcmdi.metrics: PMP Version 1.2. Zenodo. Retrieved 2021-08-16, from https://zenodo.org/record/1414560 doi: 10.5281/zenodo.1414560 Eyring, V., Bony, S., Mechl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. Geosci. Model Dev., 9(5), 1937–1958. doi: 10.5194/gmd-9.1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Sub- tropical Subsidence in Climate Sensitivity. Science, 338(6108), 792–794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and histori- cal warming evolution in recent coupled model ensembles. Atmospheric Chem- istry and Physics, 20(13), 7829–7842. Retrieved 2021-03-25, from https://aguputs operaicus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/ acp-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch,	723	Cess, R. D., Potter, G. L., Blanchet, J. P., Boer, G. J., Ghan, S. J., Kiehl, J. T.,
 Atmospheric General Circulation Models. Science, 245(4917), 513-516. doi: 10.1126/science.245.4917.513 Collins, M., Knutti, R., Arblaster, J., Dufresne, JL., Fichefet, T., Friedlingstein, P., Wehner, M. (2013). Long-term Climate Change: Projections, Commitments and Irreversibility. In Climate Change: 2013: The Physical Science Basis. Contribution of Working Group 1 to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press. Collins, W. J., Bellouin, N., Doutriaux-Boucher, M., Gedney, N., Hinton, P. H. T., Hughes, J., Woodward, S. (2011). Development and evaluation of an Earth-system model - HadGEM2. Geosci. Model Dev. Discuss., 4, 997-1062. Doutriaux, C., Gleckler, P., Durack, P. J., Lee, J., Covey, C., Sperber, K., jservonnat (2018, September). PCMDI/pcmdi.metrics: PMP Version 1.2. Zenodo. Retrieved 2021-08-16, from https://zenodo.org/record/1414560 Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. Geosci. Model Dev., 9(5), 1937-1958. doi: 10.5194/gmd-9-1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Subtropical Subsidence in Climate Sensitivity. Science, 338(6108), 792-794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and historical warming evolution in recent coupled model ensembles. Atmospheric Chemistry and Physics, 20(13), 7829-7842. Retrieved 2021-03-25, from https://agupus.on.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/ acp-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Re	724	Yagai, I. (1989). Interpretation of Cloud-Climate Feedback as Produced by 14
 10.1126/science.245.4917.513 Collins, M., Knutti, R., Arblaster, J., Dufresne, JL., Fichefet, T., Friedlingstein, P., Wehner, M. (2013). Long-term Climate Change: Projections, Commitments and Irreversibility. In Climate Change 2013: The Physical Science Basis. Contribution of Working Group 1 to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, United Kingdom and New York, NY, USA.: Cambridge University Press. Collins, W. J., Bellouin, N., Doutriaux-Boucher, M., Gedney, N., Hinton, P. H. T., Hughes, J., Woodward, S. (2011). Development and evaluation of an Earth-system model - HadGEM2. Geosci. Model Dev. Discuss., 4, 997–1062. Doutriaux, C., Gleckler, P., Durack, P. J., Lee, J., Covey, C., Sperber, K., jservonnat (2018, September). PCMDI/pendi.metrics: PMP Version 1.2. Zenodo. Retrieved 2021-08-16, from https://zenodo.org/record/1414560 Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. Geosci. Model Dev., 9(5), 1937–1958. doi: 10.5194/gmd-9-1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Subtropical Subsidence in Climate Sensitivity. Science, 338(6108), 792–794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and historical warming evolution in recent coupled model ensembles. Atmospheric Chemistry and Physics, 20(13), 7829–7842. Retrieved 2021-03-10, from https://acp.coperricus.org/articles/20/7829/2020/ doi: https://doi.org/10.194/acp-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheric Chemistry and Physics, 20(13), 7829–7842. Retrieve	725	Atmospheric General Circulation Models. Science, 245 (4917), 513–516. doi:
 Collins, M., Knutti, R., Arblaster, J., Dufresne, JL., Fichefet, T., Friedlingstein, P., Wehner, M. (2013). Long-term Climate Change: Projections, Commitments and Irreversibility. In <i>Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change.</i> Cambridge, United Kingdom and New York, NY, USA.: Cambridge University Press. Collins, W. J., Bellouin, N., Doutriaux-Boucher, M., Gedney, N., Hinton, P. H. T., Hughes, J., Woodward, S. (2011). Development and evaluation of an Earth-system model - HadGEM2. <i>Geosci. Model Dev. Discuss.</i>, <i>4</i>, 997–1062. Doutriaux, C., Gleckler, P., Durack, P. J., Lee, J., Covey, C., Sperber, K., jservonnat (2018, September). <i>PCMDI/pendi.metrics: PMP Version 1.2.</i> Zenodo. Retrieved 2021-08-16, from https://zenodo.org/record/1414560 Eyring, V., Bony, S., Mechl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. <i>Geosci. Model Dev.</i>, <i>9</i>(5), 1937–1958. doi: 10.5194/gmd-9-1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Subtropical Subsidence in Climate Sensitivity. <i>Science</i>, 338(6108), 792–794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and historical warming evolution in recent coupled model ensembles. <i>Atmospheric Chemistry and Physics</i>, 20(13), 7829–7842. Retrieved 2021-03-10, from https://acup-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. <i>Journal of Geophysical Research: Atmospheres</i>, <i>126</i>(4), e202JD033457. Retrieved 2021-03-25, from https://acupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/202DJD033487 Gent, P.	726	10.1126/science.245.4917.513
 P., Wehner, M. (2013). Long-term Climate Change: Projections, Committenents and Ireversibility. In Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press. Collins, W. J., Bellouin, N., Doutriaux-Boucher, M., Gedney, N., Hinton, P. H. T., Hughes, J., Woodward, S. (2011). Development and evaluation of an Earth-system model - HadGEM2. Geosci. Model Dev. Discuss. 4, 997-1062. Doutriaux, C., Gleckler, P., Durack, P. J., Lee, J., Covey, C., Sperber, K., jservonnat (2018, September). PCMDI/pcmdi.metrics: PMP Version 1.2. Zenodo. Retrieved 2021-08-16, from https://zenodo.org/record/1414560 doi: 10.5281/zenodo.1414560 Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. Geosci. Model Dev., 9(5), 1937-1958. doi: 10.5194/gmd-9-1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Subtropical Subsidence in Climate Sensitivity. Science, 338(6108), 792-794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and historical warming evolution in recent coupled model ensembles. Atmospheric Chemistry and Physics, 20(13), 7829-7842. Retrieved 2021-03-210, prom https://acpub.com/doi/2020JD033487 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://acpubs.onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 Gente, P. R., Danabasoglu, G., Douner, L. J., Holland, M. M., Hunke, E. C., Jay	727	Collins, M., Knutti, R., Arblaster, J., Dufresne, JL., Fichefet, T., Friedlingstein,
 mitments and Irreversibility. In Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press. Collins, W. J., Bellouin, N., Doutriaux-Boucher, M., Gedney, N., Hinton, P. H. T., Hughes, J., Woodward, S. (2011). Development and evaluation of an Earth-system model - HadGEM2. Geosci. Model Dev. Discuss., 4, 997-1062. Doutriaux, C., Gleckler, P., Durack, P. J., Lee, J., Covey, C., Sperber, K., jser- vonnat (2018, September). PCMD1/pcmdi.metrics: PMP Version 1.2. Zenodo. Retrieved 2021-08-16, from https://zenodo.org/record/1414560 doi: 10.5281/zenodo.1414560 Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. Geosci. Model Dev., 9(5), 1937-1958. doi: 10.5194/gmd-9-1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Sub- tropical Subsidence in Climate Sensitivity. Science, 338(6108), 792-794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and histori- cal warning evolution in recent coupled model ensembles. Atmospheric Chem- istry and Physics, 20(13), 7829-7842. Retrieved 2021-03-10, from https://acp .coperricus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/ acp-20-7829-0202 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs .online1ibrary.wiley.com/doi/abs/10.1029/2020JD033487 (cprint: https://doi.org/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G.,	728	P., Wehner, M. (2013). Long-term Climate Change: Projections, Com-
 Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press. Collins, W. J., Bellouin, N., Doutriaux-Boucher, M., Gedney, N., Hinton, P. H. T., Hughes, J., Woodward, S. (2011). Development and evaluation of an Earth-system model - HadGEM2. Geosci. Model Dev. Discuss., 4, 997-1062. Doutriaux, C., Gleckler, P., Durack, P. J., Lee, J., Covey, C., Sperber, K., jser- vonnat (2018, September). PCMDI/pcmdi.metrics: PMP Version 1.2. Zenodo. Retrieved 2021-08-16, from https://zenodo.org/record/1414560 doi: 10.5281/zenodo.1414560 Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. Geosci. Model Dev., 9(5), 1937-1958. doi: 10.5194/gmd-9-1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Sub- tropical Subsidence in Climate Sensitivity. Science, 338(6108), 792-794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and histori- cal warming evolution in recent coupled model ensembles. Atmospheric Chem- istry and Physics, 20(13), 7829-7842. Retrieved 2021-03-10, from https://acp .copernicus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/ acp-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs .onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 (ceprint: https://doi.org/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, s. R., Zhang, M. (729	mitments and Irreversibility. In Climate Change 2013: The Physical Science
 Intergovernmental Panel on Climate Change. Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press. Collins, W. J., Bellouin, N., Doutriaux-Boucher, M., Gedney, N., Hinton, P. H. T., Hughes, J., Woodward, S. (2011). Development and evaluation of an Earth-system model - HadGEM2. Geosci. Model Dev. Discuss., 4, 997-1062. Doutriaux, C., Gleckler, P., Durack, P. J., Lee, J., Covey, C., Sperber, K., jser- vonnat (2018, September). PCMDI/pcmdi.metrics: PMP Version 1.2. Zenodo. Retrieved 2021-08-16, from https://zenodo.org/record/1414560 Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. Geosci. Model Dev., 9(5), 1937-1958. doi: 10.5194/gmd-9-1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Sub- tropical Subsidence in Climate Sensitivity. Science, 338(6108), 792-794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and histori- cal warning evolution in recent coupled model ensembles. Atmospheric Chem- istry and Physics, 20(13), 7829-7842. Retrieved 2021-03-10, from https://acp .copernicus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/ acp-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs .onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 (ceprint: https://doi.org/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. <i>J. Climate</i>, 24, 4973-4991. doi: 10.1175/2011JCL140	730	Basis. Contribution of Working Group I to the Fifth Assessment Report of the
 New York, NY, USA.: Cambridge University Press. Collins, W. J., Bellouin, N., Doutriaux-Boucher, M., Gedney, N., Hinton, P. H. T., Hughes, J., Woodward, S. (2011). Development and evaluation of an Earth-system model - HadGEM2. Geosci. Model Dev. Discuss., 4, 997–1062. Doutriaux, C., Gleckler, P., Durack, P. J., Lee, J., Covey, C., Sperber, K., jser- vonnat (2018, September). PCMDI/pcmdi.metrics: PMP Version 1.2. Zenodo. Retrieved 2021-08-16, from https://zenodo.org/record/1414560 Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. Geosci. Model Dev., 9(5), 1937–1958. doi: 10.5194/gmd-9-1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Sub- tropical Subsidence in Climate Sensitivity. Science, 338 (6108), 792–794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and histori- cal warming evolution in recent coupled model ensembles. Atmospheric Chem- istry and Physics, 20(13), 7829–7842. Retrieved 2021-03-10, from https://acp .copernicus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/ acp-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs .onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 (ceprint: https://doi.org/10.1029/2020JD033487 (ceprint: https://doi.org/10.1029/2020JD033487 (ceprint: https://doi.org/10.1029/2020JD033487. Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973-4991. d	731	Intergovernmental Panel on Climate Change. Cambridge, United Kingdom and
 Collins, W. J., Bellouin, N., Doutriaux-Boucher, M., Gedney, N., Hinton, P. H. T., Hughes, J., Woodward, S. (2011). Development and evaluation of an Earth-system model - HadGEM2. Geosci. Model Dev. Discuss., 4, 997-1062. Doutriaux, C., Gleckler, P., Durack, P. J., Lee, J., Covey, C., Sperber, K., jser- vonnat (2018, September). PCMDI/pcmdi.metrics: PMP Version 1.2. Zenodo. Retrieved 2021-08-16, from https://zenodo.org/record/1414560 doi: 10.5281/zenodo.1414560 Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. Geosci. Model Dev., 9(5), 1937-1958. doi: 10.5194/gmd-9-1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Sub- tropical Subsidence in Climate Sensitivity. Science, 338(6108), 792-794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and histori- cal warming evolution in recent coupled model ensembles. Atmospheric Chem- istry and Physics, 20(13), 7829-7842. Retrieved 2021-03-10, from https://acp .copernicus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/ acp-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs .onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487 (e-print: https://dgupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487 doi: https://doi.org/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Onner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973-4991. doi: 10.1175/2011JCL14083.1 Gettelman, A., H	732	New York, NY, USA.: Cambridge University Press.
 Hughes, J., Woodward, S. (2011). Development and evaluation of an Earth-system model - HadGEM2. Geosci. Model Dev. Discuss., 4, 997-1062. Doutriaux, C., Gleckler, P., Durack, P. J., Lee, J., Covey, C., Sperber, K., jser- vonnat (2018, September). PCMDI/pcmdi.metrics: PMP Version 1.2. Zenodo. Retrieved 2021-08-16, from https://zenodo.org/record/1414560 doi: 10.5281/zenodo.1414560 Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. Geosci. Model Dev., 9(5), 1937-1958. doi: 10.5194/gmd-9-1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Sub- tropical Subsidence in Climate Sensitivity. Science, 338(6108), 792-794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and histori- cal warming evolution in recent coupled model ensembles. Atmospheric Chem- istry and Physics, 20(13), 7829-7842. Retrieved 2021-03-10, from https://acp .copernicus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/ acp-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973-4991. doi: 10.1175/2011JCL14083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geoph	733	Collins, W. J., Bellouin, N., Doutriaux-Boucher, M., Gedney, N., Hinton, P. H. T.,
 Earth-system model - HadGEM2. Geosci. Model Dev. Discuss., 4, 997-1062. Doutriaux, C., Gleckler, P., Durack, P. J., Lee, J., Covey, C., Sperber, K., jservonnat (2018, September). PCMDI/pcmdi.metrics: PMP Version 1.2. Zenodo. Retrieved 2021-08-16, from https://zenodo.org/record/1414560 Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. Geosci. Model Dev., 9(5), 1937-1958. doi: 10.5194/gmd-9-1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Subtropical Subsidence in Climate Sensitivity. Science, 338(6108), 792-794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and historical warming evolution in recent coupled model ensembles. Atmospheric Chemistry and Physics, 20(13), 7829-7842. Retrieved 2021-03-10, from https://acp.copernicus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/gmd-9-103-2016 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487 (.eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487 (.eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487 (.eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487 Gent, P. R., Danabasoghu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973-4991. doi: 10.1175/2011JCL14083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G.,	734	Hughes, J., Woodward, S. (2011). Development and evaluation of an
 Doutriaux, C., Gleckler, P., Durack, P. J., Lee, J., Covey, C., Sperber, K., jservonat (2018, September). PCMDI/pcmdi.metrics: PMP Version 1.2. Zenodo. Retrieved 2021-08-16, from https://zenodo.org/record/1414560 Eyring, V., Bony, S., Mechl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. Geosci. Model Dev., 9(5), 1937–1958. doi: 10.5194/gmd-9-1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Subtropical Subsidence in Climate Sensitivity. Science, 338(6108), 792–794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and historical warming evolution in recent coupled model ensembles. Atmospheric Chemistry and Physics, 20(13), 7829–7842. Retrieved 2021-03-10, from https://acp.copernicus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/acp-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs.onlinelibrary.wiley.com/doi/pdf10.1029/2020JD033487 (.eprint: https://doi.org/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973-4991. doi: 10.1175/2011JCL14083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46(14), 8329-8337. doi: 10.1029/2019g083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wo	735	Earth-system model - HadGEM2. Geosci. Model Dev. Discuss., 4, 997–1062.
 vonnat (2018, September). PCMDI/pcmdi.metrics: PMP Version 1.2. Zenodo. Retrieved 2021-08-16, from https://zenodo.org/record/1414560 doi: 10.5281/zenodo.1414560 Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. Geosci. Model Dev., 9(5), 1937–1958. doi: 10.5194/gmd-9-1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Sub- tropical Subsidence in Climate Sensitivity. Science, 338(6108), 792–794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and histori- cal warming evolution in recent coupled model ensembles. Atmospheric Chem- istry and Physics, 20(13), 7829–7842. Retrieved 2021-03-10, from https://acp .copernicus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/ acp-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs .onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 (.eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487 doi: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487 doi: https://doi.org/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973–4991. doi: 10.1175/2011JCL14083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46(14), 8329–8337. doi: 10.1029/20	736	Doutriaux, C., Gleckler, P., Durack, P. J., Lee, J., Covey, C., Sperber, K., jser-
 Retrieved 2021-08-16, from https://zenodo.org/record/1414560 doi: 10.5281/zenodo.1414560 Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. Geosci. Model Dev., 9(5), 1937–1958. doi: 10.5194/gmd-9-1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Sub- tropical Subsidence in Climate Sensitivity. Science, 338(6108), 792–794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and histori- cal warming evolution in recent coupled model ensembles. Atmospheric Chem- istry and Physics, 20(13), 7829–7842. Retrieved 2021-03-10, from https://acp .copernicus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/ acp-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs .onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 (c.eprint: https://doi.org/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973-4991. doi: 10.1175/2011JCL14083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46(14), 8329–8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standa	737	vonnat (2018, September). PCMDI/pcmdi_metrics: PMP Version 1.2. Zenodo.
 10.5281/zenodo.1414560 Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. Geosci. Model Dev., 9(5), 1937–1958. doi: 10.5194/gmd-9-1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Sub- tropical Subsidence in Climate Sensitivity. Science, 338(6108), 792–794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and histori- cal warming evolution in recent coupled model ensembles. Atmospheric Chem- istry and Physics, 20(13), 7829–7842. Retrieved 2021-03-10, from https://acp .copernicus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/ acp-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs .onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 (.eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487) doi: https://doi.org/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973–4991. doi: 10.1175/2011JCL14083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46(14), 8329–8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Re	738	Retrieved 2021-08-16, from https://zenodo.org/record/1414560 doi:
 Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. Geosci. Model Dev., 9(5), 1937–1958. doi: 10.5194/gmd-9-1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Sub- tropical Subsidence in Climate Sensitivity. Science, 338(6108), 792–794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and histori- cal warming evolution in recent coupled model ensembles. Atmospheric Chem- istry and Physics, 20(13), 7829–7842. Retrieved 2021-03-0, from https://acp .copernicus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/ acp-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs .onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 (.eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487) doi: https://doi.org/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973–4991. doi: 10.1175/2011JCL14083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46(14), 8329–8337. doi: 10.1029/2019l083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Model	739	10.5281/zenodo.1414560
 Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. Geosci. Model Dev., 9(5), 1937–1958. doi: 10.5194/gmd-9-1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Sub- tropical Subsidence in Climate Sensitivity. Science, 338(6108), 792–794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and histori- cal warming evolution in recent coupled model ensembles. Atmospheric Chem- istry and Physics, 20(13), 7829–7842. Retrieved 2021-03-10, from https://acp .copernicus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/ acp-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs .onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 (.eprint: https://doi.org/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973–4991. doi: 10.1175/2011JCL14083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46(14), 8329–8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Modeling Earth Systems, 11(7), 2089–2129. doi: 10.1029/2018ms001603 	740	Evring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., &
 Phase 6 (CMIP6) experimental design and organization. Geosci. Model Dev., 9(5), 1937–1958. doi: 10.5194/gmd-9-1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Sub- tropical Subsidence in Climate Sensitivity. Science, 338 (6108), 792–794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and histori- cal warming evolution in recent coupled model ensembles. Atmospheric Chem- istry and Physics, 20(13), 7829–7842. Retrieved 2021-03-10, from https://acp .copernicus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/ acp-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs .onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 (.eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487) doi: https://doi.org/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973-4991. doi: 10.1175/2011JCL14083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46(14), 8329-8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Modeling Earth Systems, 11(7), 2089–2129. doi: 10.1029/2018ms001603 	741	Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project
 9(5), 1937–1958. doi: 10.5194/gmd-9-1937-2016 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Subtropical Subsidence in Climate Sensitivity. Science, 338(6108), 792–794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and historical warming evolution in recent coupled model ensembles. Atmospheric Chemistry and Physics, 20(13), 7829–7842. Retrieved 2021-03-10, from https://acp.copernicus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/acp-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 (.eprint: https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487) doi: https://doi.org/10.1029/202JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973–4991. doi: 10.1175/2011JCL14083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46(14), 8329–8337. doi: 10.1029/201903978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Modeling Earth Systems, 11(7), 2089–2129. doi: 10.1029/2018ms001603 	742	Phase 6 (CMIP6) experimental design and organization. <i>Geosci. Model Dev.</i> ,
 Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Subtropical Subsidence in Climate Sensitivity. Science, 338(6108), 792–794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and historical warming evolution in recent coupled model ensembles. Atmospheric Chemistry and Physics, 20(13), 7829–7842. Retrieved 2021-03-10, from https://acp.copernicus.org/articles/20/7829/2020/ Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973–4991. doi: 10.1175/2011JCLI4083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46(14), 8329–8337. doi: 10.1029/2019J033487 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Modeling Earth Systems, 11(7), 2089–2129. doi: 10.1029/2018ms001603 	743	9(5), 1937–1958. doi: 10.5194/gmd-9-1937-2016
 tropical Subsidence in Climate Sensitivity. Science, 338(6108), 792–794. doi: 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and histori- cal warming evolution in recent coupled model ensembles. Atmospheric Chem- istry and Physics, 20(13), 7829–7842. Retrieved 2021-03-10, from https://acp .copernicus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/ acp-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs .onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487) doi: https://doi.org/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973–4991. doi: 10.1175/2011JCLI4083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46(14), 8329–8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Modeling Earth Systems, 11(7), 2089–2129. doi: 10.1029/2018ms001603 	744	Fasullo, J. T., & Trenberth, K. E. (2012). A Less Cloudy Future: The Role of Sub-
 10.1126/science.1227465 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and histori- cal warming evolution in recent coupled model ensembles. Atmospheric Chem- istry and Physics, 20(13), 7829–7842. Retrieved 2021-03-10, from https://acp .copernicus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/ acp-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs .onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487) doi: https://doi.org/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973-4991. doi: 10.1175/2011JCL14083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46(14), 8329-8337. doi: 10.1029/2019g0083078 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Modeling Earth Systems, 11(7), 2089-2129. doi: 10.1029/2018ms001603 	745	tropical Subsidence in Climate Sensitivity. <i>Science</i> , 338(6108), 792–794. doi:
 Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and histori- cal warming evolution in recent coupled model ensembles. Atmospheric Chem- istry and Physics, 20(13), 7829–7842. Retrieved 2021-03-10, from https://acp .copernicus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/ acp-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs .onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 (.eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487) doi: https://doi.org/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973–4991. doi: 10.1175/2011JCLI4083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46(14), 8329–8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Modeling Earth Systems, 11(7), 2089–2129. doi: 10.1029/2018ms001603 	746	10.1126/science.1227465
 cal warming evolution in recent coupled model ensembles. Atmospheric Chem- istry and Physics, 20(13), 7829–7842. Retrieved 2021-03-10, from https://acp copernicus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/ acp-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs .onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487) doi: https://doi.org/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973–4991. doi: 10.1175/2011JCL14083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46(14), 8329–8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Modeling Earth Systems, 11(7), 2089–2129. doi: 10.1029/2018m001603 	747	Flynn, C. M., & Mauritsen, T. (2020, July). On the climate sensitivity and histori-
 <i>istry and Physics</i>, 20(13), 7829–7842. Retrieved 2021-03-10, from https://acp .copernicus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/ acp-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, <i>126</i>(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs .onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487) Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973–4991. doi: 10.1175/2011JCLI4083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research <i>Letters</i>, 46(14), 8329–8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Modeling Earth Systems, 11(7), 2089–2129. doi: 10.1029/2018ms001603 	748	cal warming evolution in recent coupled model ensembles. Atmospheric Chem-
 .copernicus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/ acp-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126 (4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs .onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 (.eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487) doi: https://doi.org/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973-4991. doi: 10.1175/2011JCLI4083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46 (14), 8329-8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Modeling Earth Systems, 11(7), 2089-2129. doi: 10.1029/2018ms001603 	749	istry and Physics, 20(13), 7829-7842. Retrieved 2021-03-10, from https://acp
 acp-20-7829-2020 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126 (4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs .onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487) doi: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487) doi: https://doi.org/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973–4991. doi: 10.1175/2011JCLI4083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46(14), 8329–8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Modeling Earth Systems, 11(7), 2089–2129. doi: 10.1029/2018ms001603 	750	.copernicus.org/articles/20/7829/2020/ doi: https://doi.org/10.5194/
 Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021). A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs .onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487) doi: https://doi.org/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973–4991. doi: 10.1175/2011JCLI4083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46(14), 8329–8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Modeling Earth Systems, 11(7), 2089–2129. doi: 10.1029/2018ms001603 	751	acp-20-7829-2020
 A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present and Future Climate. Journal of Geophysical Research: Atmospheres, 126 (4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs .onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487) doi: https://doi.org/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973–4991. doi: 10.1175/2011JCLI4083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46(14), 8329–8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Modeling Earth Systems, 11(7), 2089–2129. doi: 10.1029/2018ms001603 	752	Gasparini, B., Rasch, P. J., Hartmann, D. L., Wall, C. J., & Dütsch, M. (2021).
 and Future Climate. Journal of Geophysical Research: Atmospheres, 126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs .onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973-4991. doi: 10.1175/2011JCL14083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46(14), 8329-8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Modeling Earth Systems, 11(7), 2089-2129. doi: 10.1029/2018ms001603 	753	A Lagrangian Perspective on Tropical Anvil Cloud Lifecycle in Present
 126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs .onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487) doi: https://doi.org/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. <i>J. Climate</i>, 24, 4973–4991. doi: 10.1175/2011JCLI4083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research <i>Letters</i>, 46(14), 8329–8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in <i>Modeling Earth Systems</i>, 11(7), 2089–2129. doi: 10.1029/2018ms001603 	754	and Future Climate. Journal of Geophysical Research: Atmospheres,
 .onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487) doi: https://doi.org/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. <i>J. Climate</i>, 24, 4973–4991. doi: 10.1175/2011JCLI4083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). <i>Geophysical Research</i> <i>Letters</i>, 46(14), 8329–8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. <i>Journal of Advances in</i> <i>Modeling Earth Systems</i>, 11(7), 2089–2129. doi: 10.1029/2018ms001603 	755	126(4), e2020JD033487. Retrieved 2021-03-25, from https://agupubs
 https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487) doi: https://doi.org/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973–4991. doi: 10.1175/2011JCLI4083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46(14), 8329–8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Modeling Earth Systems, 11(7), 2089–2129. doi: 10.1029/2018ms001603 	756	.onlinelibrary.wiley.com/doi/abs/10.1029/2020JD033487 (_eprint:
 https://doi.org/10.1029/2020JD033487 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. <i>J. Climate</i>, 24, 4973–4991. doi: 10.1175/2011JCLI4083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). <i>Geophysical Research</i> <i>Letters</i>, 46(14), 8329–8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. <i>Journal of Advances in</i> <i>Modeling Earth Systems</i>, 11(7), 2089–2129. doi: 10.1029/2018ms001603 	757	https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020JD033487) doi:
 Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., Zhang, M. (2011). The community climate system model version 4. <i>J. Climate</i>, 24, 4973–4991. doi: 10.1175/2011JCLI4083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research <i>Letters</i>, 46(14), 8329–8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in <i>Modeling Earth Systems</i>, 11(7), 2089–2129. doi: 10.1029/2018ms001603 	758	https://doi.org/10.1029/2020JD033487
 S. R., Zhang, M. (2011). The community climate system model version 4. J. Climate, 24, 4973–4991. doi: 10.1175/2011JCLI4083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46(14), 8329–8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Modeling Earth Systems, 11(7), 2089–2129. doi: 10.1029/2018ms001603 	759	Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne,
 J. Climate, 24, 4973–4991. doi: 10.1175/2011JCLI4083.1 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46(14), 8329–8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Modeling Earth Systems, 11(7), 2089–2129. doi: 10.1029/2018ms001603 	760	S. R., Zhang, M. (2011). The community climate system model version 4.
 Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G., Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46(14), 8329–8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Modeling Earth Systems, 11(7), 2089–2129. doi: 10.1029/2018ms001603 	761	J. Climate, 24, 4973–4991. doi: 10.1175/2011JCLI4083.1
 Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46 (14), 8329–8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Modeling Earth Systems, 11(7), 2089–2129. doi: 10.1029/2018ms001603 	762	Gettelman, A., Hannay, C., Bacmeister, J. T., Neale, R. B., Pendergrass, A. G.,
 Community Earth System Model Version 2 (CESM2). Geophysical Research Letters, 46(14), 8329–8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Modeling Earth Systems, 11(7), 2089–2129. doi: 10.1029/2018ms001603 	763	Danabasoglu, G., Mills, M. J. (2019). High Climate Sensitivity in the
 Letters, 46 (14), 8329–8337. doi: 10.1029/2019gl083978 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Modeling Earth Systems, 11(7), 2089–2129. doi: 10.1029/2018ms001603 	764	Community Earth System Model Version 2 (CESM2). Geophysical Research
 Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, Q., Wolfe, J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Modeling Earth Systems, 11(7), 2089–2129. doi: 10.1029/2018ms001603 	765	Letters, 46(14), 8329–8337. doi: 10.1029/2019gl083978
 J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1: Overview and Evaluation at Standard Resolution. Journal of Advances in Modeling Earth Systems, 11(7), 2089–2129. doi: 10.1029/2018ms001603 	766	Golaz, J. C., Caldwell, P. M., Van Roekel, L. P., Petersen, M. R., Tang, O., Wolfe
768Overview and Evaluation at Standard Resolution.Journal of Advances in769Modeling Earth Systems, 11(7), 2089–2129. doi: 10.1029/2018ms001603	767	J. D., Zhu, Q. (2019). The DOE E3SM Coupled Model Version 1:
Modeling Earth Systems, 11(7), 2089-2129. doi: 10.1029/2018ms001603	768	Overview and Evaluation at Standard Resolution. Journal of Advances in
	769	Modeling Earth Systems, 11(7), 2089–2129. doi: 10.1029/2018ms001603
⁷⁷⁰ Golaz, JC., Horowitz, L. W., & Levy, H. (2013). Cloud tuning in a cou-	770	Golaz, JC., Horowitz, L. W., & Levy, H. (2013). Cloud tuning in a cou-

771	pled climate model: Impact on 20th century warming. Geophysical Re-
772	search Letters, 40(10), 2246–2251. Retrieved 2021-07-24, from https://
773	agupubs.onlinelibrary.wiley.com/doi/abs/10.1002/grl.50232 (_eprint:
774	https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1002/grl.50232) doi:
775	$10.1002/ m{grl}.50232$
776	Hajima, T., Watanabe, M., Yamamoto, A., Tatebe, H., Noguchi, M. A., Abe, M.,
777	Kawamiya, M. (2020, May). Development of the MIROC-ES2L Earth
778	system model and the evaluation of biogeochemical processes and feedbacks.
779	Geoscientific Model Development, 13(5), 2197–2244. Retrieved 2021-07-17,
780	from https://gmd.copernicus.org/articles/13/2197/2020/ (Publisher:
781	Copernicus GmbH) doi: $10.5194/gmd-13-2197-2020$
782	Hartmann, D. L. (2016, August). Tropical anvil clouds and climate sensitivity.
783	Proceedings of the National Academy of Sciences. Retrieved 2021-03-25,
784	$\mathrm{from}\ \mathtt{https://www.pnas.org/content/early/2016/07/29/1610455113}$
785	(Publisher: National Academy of Sciences Section: Commentary) doi:
786	10.1073/pnas.1610455113
787	Held, I. M., Guo, H., Adcroft, A., Dunne, J. P., Horowitz, L. W., Krasting,
788	J., Zadeh, N. (2019). Structure and Performance of GFDL's
789	CM4.0 Climate Model. Journal of Advances in Modeling Earth Sys-
790	tems, 11(11), 3691–3727. Retrieved 2021-07-17, from https://agupubs
791	.onlinelibrary.wiley.com/doi/abs/10.1029/2019MS001829 (_eprint:
792	https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2019MS001829) doi:
793	10.1029/2019MS001829
794	Huang, Y., Xia, Y., & Tan, X. X. (2017). On the pattern of CO2 radiative forcing
795	and poleward energy transport. Journal of Geophysical Research-Atmospheres,
796	122(20), 10578–10593. doi: 10.1002/2017jd027221
797	Jing, X., & Suzuki, K. (2018). The Impact of Process-Based Warm Rain Con-
798	straints on the Aerosol Indirect Effect. <i>Geophysical Research Letters</i> ,
799	45(19), 10,729–10,737. Retrieved 2021-07-26, from https://agupubs
800	.onlinelibrary.wiley.com/doi/abs/10.1029/2018GL079956 (_eprint:
801	https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2018GL079956) doi:
802	10.1029/2018GL079956
803	Klein, S. A., Hall, A., Norris, J. R., & Pincus, R. (2017). Low-Cloud Feedbacks from
804	Cloud-Controlling Factors: A Review. Surveys in Geophysics. doi: 10.1007/
805	s10712-017-9433-3
806	Klein, S. A., & Jakob, C. (1999). Validation and sensitivities of frontal clouds sim-
807	ulated by the ECMWF model. Mon. Weath. Rev., 127, 2514–2531. doi: 10
808	.1175/1520-0493(1999)1272.0.CO;2
809	Klein, S. A., Zhang, Y., Zelinka, M. D., Pincus, R., Boyle, J., & Gleckler, P. J.
810	(2013). Are climate model simulations of clouds improving? An evaluation
811	using the ISCCP simulator. Journal of Geophysical Research-Atmospheres,
812	118(3), 1329–1342. doi: 10.1002/jgrd.50141
813	Mauritsen, T., & Stevens, B. (2015). Missing iris effect as a possible cause of muted
814	hydrological change and high climate sensitivity in models. Nature Geosci,
815	8(5), 346–351. doi: 10.1038/ngeo2414http://www.nature.com/ngeo/journal/
816	v8/n5/abs/ngeo2414.html#supplementary-information
817	Myers, T. A., Scott, R. C., Zelinka, M. D., Klein, S. A., Norris, J. R., & Cald-
818	well, P. M. (2021, June). Observational constraints on low cloud feedback
819	reduce uncertainty of climate sensitivity. Nature Climate Change, 11(6).
820	501-507. Retrieved 2021-07-12, from https://www.nature.com/articles/
821	s41558-021-01039-0 (Number: 6 Publisher: Nature Publishing Group) doi:
822	10.1038/s41558-021-01039-0
823	Mülmenstädt, J., Salzmann, M., Kay, J. E., Zelinka, M. D., Ma, PL., Nam. C
824	Quaas, J. (2021, June). An underestimated negative cloud feedback from cloud
825	lifetime changes. Nature Climate Change, 11(6), 508–513. Retrieved 2021-07-

826	24, from https://www.nature.com/articles/s41558-021-01038-1 (Num-
827	ber: 6 Publisher: Nature Publishing Group) doi: 10.1038/s41558-021-01038-1
828	Nijsse, F. J. M. M., Cox, P. M., & Williamson, M. S. (2020, August). Emer-
829	gent constraints on transient climate response (TCR) and equilibrium cli-
830	mate sensitivity (ECS) from historical warming in CMIP5 and CMIP6 mod-
831	els. Earth System Dynamics, 11(3), 737–750. Retrieved 2021-03-19, from
832	https://esd.copernicus.org/articles/11/737/2020/ (Publisher: Coper-
833	nicus GmbH) doi: https://doi.org/10.5194/esd-11-737-2020
834	Pendergrass, A. G., Conley, A., & Vitt, F. M. (2018). Surface and top-of-atmosphere
835	radiative feedback kernels for CESM-CAM5. Earth System Science Data,
836	10(1), 317-324. doi: $10.5194/essd-10-317-2018$
837	Pincus, R., Forster, P. M., & Stevens, B. (2016). The Radiative Forcing Model In-
838	tercomparison Project (RFMIP): experimental protocol for CMIP6. Geoscien-
839	tific Model Development, 9(9), 3447–3460. doi: 10.5194/gmd-9-3447-2016
840	Ringer, M. A., Andrews, T., & Webb, M. J. (2014). Global-mean radiative feed-
841	backs and forcing in atmosphere-only and coupled atmosphere-ocean climate
842	change experiments. <i>Geophysical Research Letters</i> , 41(11), 4035–4042. doi:
843	10.1002/2014gl 060347
844	Rossow, W. B., & Schiffer, R. A. (1999). Advances in Understanding Clouds
845	from ISCCP. Bull. Amer. Meteor. Soc., 80(11), 2261–2287. doi: 10.1175/
846	1520-0477(1999)0802.0.CO:2
947	Scott B C Myers T A Norris J B Zelinka M D Klein S A Sun M
848	& Doelling D B (2020 September) Observed Sensitivity of Low-Cloud
940	Badiative Effects to Meteorological Perturbations over the Global Oceans
950	Journal of Climate 33(18) 7717–7734 Retrieved 2021-04-21 from https://
951	journals ametsoc org/view/journals/clim/33/18/icliD191028 xml
051	(Publisher: American Meteorological Society Section: Journal of Climate) doi:
052	(i ubisite i interest interest bisgical bellety beston. bound of children) as:
952	10 1175/JCLI-D-19-1028 1
853	10.1175/JCLI-D-19-1028.1 Seeley J. T. Jeevaniee, N. Langhans, W. & Romps, D. M. (2019) Formation
853 854	10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation Geophysical Research Letters
853 854 855	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. <i>Geophysical Research Letters</i>, <i>16</i>(1), 492–501. doi: 10.1029/2018GL080747
853 854 855 856	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. <i>Geophysical Research Letters</i>, 46(1), 492–501. doi: 10.1029/2018GL080747 Sellar A A Jones C G Mulcaby J P Tang Y Yool A Wiltshire A
853 854 855 856 857	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. <i>Geophysical Research Letters</i>, 46(1), 492–501. doi: 10.1029/2018GL080747 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., Zerroukat, M. (2019). UKESM1: Description and Evaluation of the
853 854 855 856 857 858	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. Geophysical Research Letters, 46(1), 492–501. doi: 10.1029/2018GL080747 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., Zerroukat, M. (2019). UKESM1: Description and Evaluation of the U.K. Earth System Model Journal of Advances in Modeling Earth System
853 854 855 856 857 858 858 859	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. Geophysical Research Letters, 46(1), 492–501. doi: 10.1029/2018GL080747 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., Zerroukat, M. (2019). UKESM1: Description and Evaluation of the U.K. Earth System Model. Journal of Advances in Modeling Earth Systems 11(12) 4513–4558 Retrieved 2021-07-17 from https://agupubs
853 854 855 856 857 858 859 860 861	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. Geophysical Research Letters, 46(1), 492–501. doi: 10.1029/2018GL080747 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., Zerroukat, M. (2019). UKESM1: Description and Evaluation of the U.K. Earth System Model. Journal of Advances in Modeling Earth Systems, 11(12), 4513–4558. Retrieved 2021-07-17, from https://agupubs.conlinelibrary.wiley.com/doi/abs/10.1029/2019MS001739 (eprint:
853 854 855 856 857 858 859 860 861 862	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. Geophysical Research Letters, 46(1), 492–501. doi: 10.1029/2018GL080747 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., Zerroukat, M. (2019). UKESM1: Description and Evaluation of the U.K. Earth System Model. Journal of Advances in Modeling Earth Systems, 11(12), 4513–4558. Retrieved 2021-07-17, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019MS001739 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019MS001739) doi:
853 854 855 856 857 858 859 860 861 862 863	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. Geophysical Research Letters, 46(1), 492–501. doi: 10.1029/2018GL080747 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., Zerroukat, M. (2019). UKESM1: Description and Evaluation of the U.K. Earth System Model. Journal of Advances in Modeling Earth Systems, 11(12), 4513–4558. Retrieved 2021-07-17, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019MS001739 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2019MS001739) doi: 10.1029/2019MS001739
853 854 855 856 857 858 859 860 861 862 863	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. Geophysical Research Letters, 46(1), 492-501. doi: 10.1029/2018GL080747 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., Zerroukat, M. (2019). UKESM1: Description and Evaluation of the U.K. Earth System Model. Journal of Advances in Modeling Earth Systems, 11(12), 4513-4558. Retrieved 2021-07-17, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019MS001739 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2019MS001739) doi: 10.1029/2019MS001739
853 854 855 856 857 858 859 860 861 862 863 864	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. Geophysical Research Letters, 46(1), 492–501. doi: 10.1029/2018GL080747 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., Zerroukat, M. (2019). UKESM1: Description and Evaluation of the U.K. Earth System Model. Journal of Advances in Modeling Earth Systems, 11(12), 4513–4558. Retrieved 2021-07-17, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019MS001739 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2019MS001739) doi: 10.1029/2019MS001739 Shell, K. M., Kiehl, J. T., & Shields, C. A. (2008). Using the Radiative Kernel Technique to Calculate Climate Feedbacks in NCAR's Community Atmospheric
853 854 855 856 857 858 860 861 862 863 864 865	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. Geophysical Research Letters, 46(1), 492-501. doi: 10.1029/2018GL080747 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., Zerroukat, M. (2019). UKESM1: Description and Evaluation of the U.K. Earth System Model. Journal of Advances in Modeling Earth Systems, 11(12), 4513-4558. Retrieved 2021-07-17, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019MS001739 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2019MS001739) doi: 10.1029/2019MS001739 Shell, K. M., Kiehl, J. T., & Shields, C. A. (2008). Using the Radiative Kernel Technique to Calculate Climate Feedbacks in NCAR's Community Atmospheric Model. J. Climate. 21(10), 2269-2282. doi: 10.1175/2007JCLI2044.1
853 854 855 856 857 858 860 861 862 863 864 865 866	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. Geophysical Research Letters, 46(1), 492-501. doi: 10.1029/2018GL080747 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., Zerroukat, M. (2019). UKESM1: Description and Evaluation of the U.K. Earth System Model. Journal of Advances in Modeling Earth Systems, 11(12), 4513-4558. Retrieved 2021-07-17, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019MS001739 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2019MS001739) doi: 10.1029/2019MS001739 Shell, K. M., Kiehl, J. T., & Shields, C. A. (2008). Using the Radiative Kernel Technique to Calculate Climate Feedbacks in NCAR's Community Atmospheric Model. J. Climate, 21(10), 2269-2282. doi: 10.1175/2007JCLI2044.1 Sherwood, S. C. Bony, S. & Dufreno, J. L. (2014). Spread in model climate sensi.
853 854 855 856 857 858 859 860 861 862 863 864 865 866	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. Geophysical Research Letters, 46(1), 492-501. doi: 10.1029/2018GL080747 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., Zerroukat, M. (2019). UKESM1: Description and Evaluation of the U.K. Earth System Model. Journal of Advances in Modeling Earth Systems, 11(12), 4513-4558. Retrieved 2021-07-17, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019MS001739 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2019MS001739) doi: 10.1029/2019MS001739 Shell, K. M., Kiehl, J. T., & Shields, C. A. (2008). Using the Radiative Kernel Technique to Calculate Climate Feedbacks in NCAR's Community Atmospheric Model. J. Climate, 21(10), 2269-2282. doi: 10.1175/2007JCLI2044.1 Sherwood, S. C., Bony, S., & Dufresne, JL. (2014). Spread in model climate sensitivity traged to atmospheric convertive mixing. Nature 505(7481), 37-42. doi:
853 854 855 856 857 858 859 860 861 862 863 864 865 866 867	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. Geophysical Research Letters, 46(1), 492–501. doi: 10.1029/2018GL080747 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., Zerroukat, M. (2019). UKESM1: Description and Evaluation of the U.K. Earth System Model. Journal of Advances in Modeling Earth Systems, 11(12), 4513–4558. Retrieved 2021-07-17, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019MS001739 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2019MS001739) doi: 10.1029/2019MS001739 Shell, K. M., Kiehl, J. T., & Shields, C. A. (2008). Using the Radiative Kernel Technique to Calculate Climate Feedbacks in NCAR's Community Atmospheric Model. J. Climate, 21(10), 2269–2282. doi: 10.1175/2007JCLI2044.1 Sherwood, S. C., Bony, S., & Dufresne, JL. (2014). Spread in model climate sensitivity traced to atmospheric convective mixing. Nature, 505(7481), 37–42. doi: 10.1038/nature12829
853 854 855 856 857 858 859 860 861 862 863 864 865 866 866 868	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. Geophysical Research Letters, 46(1), 492-501. doi: 10.1029/2018GL080747 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., Zerroukat, M. (2019). UKESM1: Description and Evaluation of the U.K. Earth System Model. Journal of Advances in Modeling Earth Systems, 11(12), 4513-4558. Retrieved 2021-07-17, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019MS001739 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2019MS001739) doi: 10.1029/2019MS001739 Shell, K. M., Kiehl, J. T., & Shields, C. A. (2008). Using the Radiative Kernel Technique to Calculate Climate Feedbacks in NCAR's Community Atmospheric Model. J. Climate, 21(10), 2269-2282. doi: 10.1175/2007JCLI2044.1 Sherwood, S. C., Bony, S., & Dufresne, JL. (2014). Spread in model climate sensitivity traced to atmospheric convective mixing. Nature, 505(7481), 37-42. doi: 10.1038/nature12829
 853 854 855 857 858 859 860 861 862 863 864 865 866 867 868 869 870 	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. Geophysical Research Letters, 46(1), 492–501. doi: 10.1029/2018GL080747 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., Zerroukat, M. (2019). UKESM1: Description and Evaluation of the U.K. Earth System Model. Journal of Advances in Modeling Earth Systems, 11(12), 4513–4558. Retrieved 2021-07-17, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019MS001739 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2019MS001739) doi: 10.1029/2019MS001739 Shell, K. M., Kiehl, J. T., & Shields, C. A. (2008). Using the Radiative Kernel Technique to Calculate Climate Feedbacks in NCAR's Community Atmospheric Model. J. Climate, 21(10), 2269–2282. doi: 10.1175/2007JCLI2044.1 Sherwood, S. C., Bony, S., & Dufresne, JL. (2014). Spread in model climate sensitivity traced to atmospheric convective mixing. Nature, 505(7481), 37–42. doi: 10.1038/nature12829 Sherwood, S. C., Webb, M. J., Annan, J. D., Armour, K. C., Forster, P. M., Harmoury, L C. (2012) An Academic Aca
853 854 855 856 857 858 860 861 862 863 864 865 866 866 866 866 866 866 866 870 871	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. Geophysical Research Letters, 46(1), 492–501. doi: 10.1029/2018GL080747 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., Zerroukat, M. (2019). UKESM1: Description and Evaluation of the U.K. Earth System Model. Journal of Advances in Modeling Earth Systems, 11(12), 4513–4558. Retrieved 2021-07-17, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019MS001739 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2019MS001739) doi: 10.1029/2019MS001739 Shell, K. M., Kiehl, J. T., & Shields, C. A. (2008). Using the Radiative Kernel Technique to Calculate Climate Feedbacks in NCAR's Community Atmospheric Model. J. Climate, 21(10), 2269–2282. doi: 10.1175/2007JCLI2044.1 Sherwood, S. C., Bony, S., & Dufresne, JL. (2014). Spread in model climate sensitivity traced to atmospheric convective mixing. Nature, 505(7481), 37–42. doi: 10.1038/nature12829 Sherwood, S. C., Webb, M. J., Annan, J. D., Armour, K. C., Forster, P. M., Hargreaves, J. C., Zelinka, M. D. (2020). An Assessment of Earth's Climate Sensitivity traced for an and sensitivity traced for a formation of the sensitivity traced formation of the sensitivity traced for a formation of the sensitivity traced formation of the sensitivi
 853 854 855 856 857 858 859 860 861 862 863 866 866 867 868 869 871 872 	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. Geophysical Research Letters, 46(1), 492-501. doi: 10.1029/2018GL080747 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., Zerroukat, M. (2019). UKESM1: Description and Evaluation of the U.K. Earth System Model. Journal of Advances in Modeling Earth Systems, 11(12), 4513-4558. Retrieved 2021-07-17, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019MS001739 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2019MS001739) doi: 10.1029/2019MS001739 Shell, K. M., Kiehl, J. T., & Shields, C. A. (2008). Using the Radiative Kernel Technique to Calculate Climate Feedbacks in NCAR's Community Atmospheric Model. J. Climate, 21(10), 2269-2282. doi: 10.1175/2007JCLI2044.1 Sherwood, S. C., Bony, S., & Dufresne, JL. (2014). Spread in model climate sensitivity traced to atmospheric convective mixing. Nature, 505(7481), 37-42. doi: 10.1038/nature12829 Sherwood, S. C., Webb, M. J., Annan, J. D., Armour, K. C., Forster, P. M., Hargreaves, J. C., Zelinka, M. D. (2020). An Assessment of Earth's Climate Sensitivity Using Multiple Lines of Evidence. Reviews of Geombusics 58(4), e2019BC000678
 853 854 855 856 857 858 859 860 861 862 863 866 866 867 868 869 870 871 872 873 	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. Geophysical Research Letters, 46(1), 492-501. doi: 10.1029/2018GL080747 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., Zerroukat, M. (2019). UKESM1: Description and Evaluation of the U.K. Earth System Model. Journal of Advances in Modeling Earth Systems, 11(12), 4513-4558. Retrieved 2021-07-17, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019MS001739 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2019MS001739) doi: 10.1029/2019MS001739 Shell, K. M., Kiehl, J. T., & Shields, C. A. (2008). Using the Radiative Kernel Technique to Calculate Climate Feedbacks in NCAR's Community Atmospheric Model. J. Climate, 21(10), 2269-2282. doi: 10.1175/2007JCLI2044.1 Sherwood, S. C., Bony, S., & Dufresne, JL. (2014). Spread in model climate sensitivity traced to atmospheric convective mixing. Nature, 505(7481), 37-42. doi: 10.1038/nature12829 Sherwood, S. C., Webb, M. J., Annan, J. D., Armour, K. C., Forster, P. M., Hargreaves, J. C., Zelinka, M. D. (2020). An Assessment of Earth's Climate Sensitivity Using Multiple Lines of Evidence. Reviews of Geophysics, 58(4), e2019RG000678. Retrieved 2021-02-14, from https://agupubs.com/aduitable
 853 854 855 856 857 858 859 860 861 862 863 864 865 866 867 868 869 870 871 872 873 874 	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. Geophysical Research Letters, 46(1), 492–501. doi: 10.1029/2018GL080747 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., Zerroukat, M. (2019). UKESM1: Description and Evaluation of the U.K. Earth System Model. Journal of Advances in Modeling Earth Systems, 11(12), 4513–4558. Retrieved 2021-07-17, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019MS001739 (.eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2019MS001739) doi: 10.1029/2019MS001739 Shell, K. M., Kiehl, J. T., & Shields, C. A. (2008). Using the Radiative Kernel Technique to Calculate Climate Feedbacks in NCAR's Community Atmospheric Model. J. Climate, 21(10), 2269–2282. doi: 10.1175/2007JCLI2044.1 Sherwood, S. C., Bony, S., & Dufresne, JL. (2014). Spread in model climate sensitivity traced to atmospheric convective mixing. Nature, 505(7481), 37–42. doi: 10.1038/nature12829 Sherwood, S. C., Webb, M. J., Annan, J. D., Armour, K. C., Forster, P. M., Hargreaves, J. C., Zelinka, M. D. (2020). An Assessment of Earth's Climate Sensitivity Using Multiple Lines of Evidence. Reviews of Geophysics, 58(4), e2019RG000678. Retrieved 2021-02-14, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019RG000678 doi: https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019RG00067
 853 854 855 856 859 860 861 862 863 864 865 866 867 868 869 870 871 872 873 874 	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. Geophysical Research Letters, 46(1), 492-501. doi: 10.1029/2018GL080747 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., Zerroukat, M. (2019). UKESM1: Description and Evaluation of the U.K. Earth System Model. Journal of Advances in Modeling Earth Systems, 11(12), 4513-4558. Retrieved 2021-07-17, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019MS001739 (.eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2019MS001739) doi: 10.1029/2019MS001739 Shell, K. M., Kiehl, J. T., & Shields, C. A. (2008). Using the Radiative Kernel Technique to Calculate Climate Feedbacks in NCAR's Community Atmospheric Model. J. Climate, 21(10), 2269-2282. doi: 10.1175/2007JCLI2044.1 Sherwood, S. C., Bony, S., & Dufresne, JL. (2014). Spread in model climate sensitivity traced to atmospheric convective mixing. Nature, 505(7481), 37-42. doi: 10.1038/nature12829 Sherwood, S. C., Webb, M. J., Annan, J. D., Armour, K. C., Forster, P. M., Hargreaves, J. C., Zelinka, M. D. (2020). An Assessment of Earth's Climate Sensitivity Using Multiple Lines of Evidence. Reviews of Geophysics, 58(4), e2019RG000678. Retrieved 2021-02-14, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019RG000678 doi: https://doi.org/10.1029/2019RG000678
 853 854 855 856 857 858 859 860 861 862 863 864 865 866 867 868 869 870 871 872 873 874 875 876 	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. Geophysical Research Letters, 46(1), 492–501. doi: 10.1029/2018GL080747 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., Zerroukat, M. (2019). UKESM1: Description and Evaluation of the U.K. Earth System Model. Journal of Advances in Modeling Earth Systems, 11(12), 4513–4558. Retrieved 2021-07-17, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019MS001739 (.eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2019MS001739) doi: 10.1029/2019MS001739 Shell, K. M., Kiehl, J. T., & Shields, C. A. (2008). Using the Radiative Kernel Technique to Calculate Climate Feedbacks in NCAR's Community Atmospheric Model. J. Climate, 21(10), 2269–2282. doi: 10.1175/2007JCLI2044.1 Sherwood, S. C., Bony, S., & Dufresne, JL. (2014). Spread in model climate sensitivity traced to atmospheric convective mixing. Nature, 505(7481), 37–42. doi: 10.1038/nature12829 Sherwood, S. C., Webb, M. J., Annan, J. D., Armour, K. C., Forster, P. M., Hargreaves, J. C., Zelinka, M. D. (2020). An Assessment of Earth's Climate Sensitivity Using Multiple Lines of Evidence. Reviews of Geophysics, 58(4), e2019RG000678. Retrieved 2021-02-14, from https:// agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019RG000678 doi: https://doi.org/10.1029/2019RG000678 Siler, N., Po-Chedley, S., & Bretherton, C. S. (2018). Variability in modeled cloud for addition of the sensitivity in the adimetion of the sensitivity in the dimetological metric location of feature of shurds. Climate feature of the sensitivity in the climate field pattern of shurds. Climate feature of shurds. Climate f
 853 854 855 856 857 858 859 860 861 862 863 866 867 868 869 870 871 872 873 874 875 876 877 	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. Geophysical Research Letters, 46(1), 492-501. doi: 10.1029/2018GL080747 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., Zerroukat, M. (2019). UKESM1: Description and Evaluation of the U.K. Earth System Model. Journal of Advances in Modeling Earth Systems, 11(12), 4513-4558. Retrieved 2021-07-17, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019MS001739 (.eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2019MS001739) doi: 10.1029/2019MS001739 Shell, K. M., Kiehl, J. T., & Shields, C. A. (2008). Using the Radiative Kernel Technique to Calculate Climate Feedbacks in NCAR's Community Atmospheric Model. J. Climate, 21(10), 2269-2282. doi: 10.1175/2007JCLI2044.1 Sherwood, S. C., Bony, S., & Dufresne, JL. (2014). Spread in model climate sensitivity traced to atmospheric convective mixing. Nature, 505(7481), 37-42. doi: 10.1038/nature12829 Sherwood, S. C., Webb, M. J., Annan, J. D., Armour, K. C., Forster, P. M., Hargreaves, J. C., Zelinka, M. D. (2020). An Assessment of Earth's Climate Sensitivity Using Multiple Lines of Evidence. Reviews of Geophysics, 58(4), e2019RG000678. Retrieved 2021-02-14, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019RG000678 doi: https://doi.org/10.1029/2019RG000678 Siler, N., Po-Chedley, S., & Bretherton, C. S. (2018). Variability in modeled cloud feedback tied to differences in the climatological spatial pattern of clouds. Climate Approximate 50(2), 1200. 1220. doi: 10.1027/c00292.0172.22
 853 854 855 856 857 858 859 860 861 862 866 866 867 871 872 873 874 875 876 877 878 	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. Geophysical Research Letters, 46(1), 492-501. doi: 10.1029/2018GL080747 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., Zerroukat, M. (2019). UKESMI: Description and Evaluation of the U.K. Earth System Model. Journal of Advances in Modeling Earth Systems, 11(12), 4513-4558. Retrieved 2021-07-17, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019MS001739 (.eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2019MS001739) doi: 10.1029/2019MS001739 Shell, K. M., Kiehl, J. T., & Shields, C. A. (2008). Using the Radiative Kernel Technique to Calculate Climate Feedbacks in NCAR's Community Atmospheric Model. J. Climate, 21(10), 2269-2282. doi: 10.1175/2007JCLI2044.1 Sherwood, S. C., Bony, S., & Dufresne, JL. (2014). Spread in model climate sensitivity traced to atmospheric convective mixing. Nature, 505(7481), 37-42. doi: 10.1038/nature12829 Sherwood, S. C., Webb, M. J., Annan, J. D., Armour, K. C., Forster, P. M., Hargreaves, J. C., Zelinka, M. D. (2020). An Assessment of Earth's Climate Sensitivity Using Multiple Lines of Evidence. Reviews of Geophysics, 58(4), e2019RG000678. Retrieved 2021-02-14, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019RG000678 doi: https://doi.org/10.1029/2019RG000678 Siler, N., Po-Chedley, S., & Bretherton, C. S. (2018). Variability in modeled cloud feedback tied to differences in the climatological spatial pattern of clouds. Climate Dynamics, 50(3), 1209-1220. doi: 10.107/s00382-017-3673-2
 853 854 855 856 857 858 859 860 861 862 863 866 866 867 868 869 870 871 872 873 874 875 876 877 878 879 	 10.1175/JCLI-D-19-1028.1 Seeley, J. T., Jeevanjee, N., Langhans, W., & Romps, D. M. (2019). Formation of Tropical Anvil Clouds by Slow Evaporation. Geophysical Research Letters, 46(1), 492-501. doi: 10.1029/2018GL080747 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., Zerroukat, M. (2019). UKESM1: Description and Evaluation of the U.K. Earth System Model. Journal of Advances in Modeling Earth Systems, 11(12), 4513-4558. Retrieved 2021-07-17, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019MS001739 (_eprint: https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2019MS001739) doi: 10.1029/2019MS001739 Shell, K. M., Kiehl, J. T., & Shields, C. A. (2008). Using the Radiative Kernel Technique to Calculate Climate Feedbacks in NCAR's Community Atmospheric Model. J. Climate, 21(10), 2269-2282. doi: 10.1175/2007JCLI2044.1 Sherwood, S. C., Bony, S., & Dufresne, JL. (2014). Spread in model climate sensitivity traced to atmospheric convective mixing. Nature, 505(7481), 37-42. doi: 10.1038/nature12829 Sherwood, S. C., Webb, M. J., Annan, J. D., Armour, K. C., Forster, P. M., Hargreaves, J. C., Zelinka, M. D. (2020). An Assessment of Earth's Climate Sensitivity Using Multiple Lines of Evidence. Reviews of Geophysics, 58(4), e2019RG000678. Retrieved 2021-02-14, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2019RG000678 doi: https://doi.org/10.1029/2019RG000678 Siler, N., Po-Chedley, S., & Bretherton, C. S. (2018). Variability in modeled cloud feedback tied to differences in the climatological spatial pattern of clouds. Climate Dynamics, 50(3), 1209-1220. doi: 10.1007/s00382-017-3673-2 Smith, C. J., Kramer, R. J., Myhre, G., Forster, P. M., Soden, B. J., Andrews, T., Watcon Rearies, 0. (2019). Understand additionate to Divense the Divense of the Dynamics. Advances of David. Divense David.

881 882	Forcing Agents. Geophysical Research Letters, $45(21)$, 12023 – 12031 . doi: $10.1029/2018$ gl079826
883	Soden, B. J., Held, I. M., Colman, R., Shell, K. M., Kiehl, J. T., & Shields, C. A.
884	(2008). Quantifying Climate Feedbacks Using Radiative Kernels. J. Climate.
885	21, 3504–3520. doi: 10.1175/2007JCLI2110.1
886	Stevens, B., Giorgetta, M., Esch, M., Mauritsen, T., Crueger, T., Rast, S.,
887	Roeckner, E. (2013). Atmospheric component of the MPI-M Earth System
888	Model: ECHAM6. Journal of Advances in Modelina Earth Systems, 5(2).
889	146–172. doi: 10.1002/jame.20015
890	Suzuki, K., Golaz, JC., & Stephens, G. L. (2013). Evaluating cloud tun-
891	ing in a climate model with satellite observations. <i>Geophysical Re-</i>
892	search Letters, 40(16), 4464–4468. Retrieved 2021-07-26, from https://
893	agupubs.onlinelibrary.wiley.com/doi/abs/10.1002/grl.50874 (_eprint:
894	https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1002/grl.50874) doi:
895	10.1002/grl.50874
896	Swart, N. C., Cole, J. N. S., Kharin, V. V., Lazare, M., Scinocca, J. F., Gillett,
897	N. P., Winter, B. (2019, November). The Canadian Earth System
898	Model version 5 (CanESM5.0.3). Geoscientific Model Development, 12(11),
899	4823-4873. Retrieved 2021-07-17, from https://gmd.copernicus.org/
900	articles/12/4823/2019/ (Publisher: Copernicus GmbH) doi: 10.5194/
901	gmd-12-4823-2019
902	Séférian, R., Nabat, P., Michou, M., Saint-Martin, D., Voldoire, A., Colin, J.,
903	Madec, G. (2019). Evaluation of CNRM Earth System Model,
904	CNRM-ESM2-1: Role of Earth System Processes in Present-Day and
905	Future Climate. Journal of Advances in Modeling Earth Systems,
906	11(12), 4182–4227. Retrieved 2021-07-17, from https://agupubs
907	.onlinelibrary.wiley.com/doi/abs/10.1029/2019MS001791 (_eprint:
908	eq:https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2019MS001791) doi:
909	10.1029/2019MS001791
910	Tatebe, H., Ogura, T., Nitta, T., Komuro, Y., Ogochi, K., Takemura, T., Ki-
911	moto, M. (2019, July). Description and basic evaluation of simulated mean
912	state, internal variability, and climate sensitivity in MIROC6. Geoscien-
913	<i>tific Model Development</i> , 12(7), 2727–2765. Retrieved 2021-07-17, from
914	https://gmd.copernicus.org/articles/12/2727/2019/ (Publisher: Coper-
915	micus GmbH) doi: $10.5194/\text{gmd}-12-2727-2019$
916	Taylor, K. E., Crucifix, M., Braconnot, P., Hewitt, C. D., Doutriaux, C., Broc-
917	coli, A. J., Webb, M. J. (2007). Estimating Shortwave Radiative Forc-
918	Ing and Response in Climate Models. J. Climate, $20(11)$, 2530–2543. doi: 10.1175 (JCH 14142.1
919	10.11(0) J U L 14 14 3.1
920	Itan, B. (2015). Spread of model climate sensitivity linked to double-
921	(2(10) 4122 4141 Detrieved 2021 08 04 from https://organical.com/
922	42(10), 4155-4141. Retrieved 2021-08-04, from https://agupubs
923	bttps://acupuba.onlinelibrory.wiley.com/doi/abs/10.1002/2015GL064119 (_eprint:
924	10,1002/2015GL064119 doi:
925	Takaraka K P. Stolno M P. Sinnol S. Fischer F. M. Smith C. I. Lahner F.
926	lokalska, K. D., Stolpe, M. D., Sippel, S., Fischer, E. M., Shifti, C. J., Lenner, F.,
927	in CMIP6 models Science Advances $6(12)$ eps29540 Betrieved 2021
920	$13-19$ from https://advances_sciencemag_org/content/6/12/epag0540
930	(Publisher: American Association for the Advancement of Science Section)
931	Research Article) doi: 10.1126/sciady.aaz9549
932	Trenberth, K. E., & Fasullo, J. T. (2010). Simulation of Present-Day and Twenty-
933	First-Century Energy Budgets of the Southern Oceans. J. Climate 23(2)
934	440–454. doi: 10.1175/2009JCLI3152.1
935	Tsushima, Y., Ringer, M. A., Martin, G. M., Rostron, J. W., & Sexton, D. M. H.

936	(2020, September). Investigating physical constraints on climate feedbacks
937	using a perturbed parameter ensemble. <i>Climate Dynamics</i> , 55(5), 1159–1185.
938	Retrieved 2021-07-24, from https://doi.org/10.1007/s00382-020-05318-y
939	doi: 10.1007/s00382-020-05318-y
940	Voldoire, A., Saint-Martin, D., Sénési, S., Decharme, B., Alias, A., Chevallier,
941	M Waldman, R. (2019). Evaluation of CMIP6 DECK Experi-
942	ments With CNRM-CM6-1. Journal of Advances in Modelina Earth Sus-
0/3	tems $11(7)$ $2177-2213$ Retrieved $2021-07-17$ from https://agupubs
044	onlinelibrary wiley com/doi/abs/10 1029/2019MS001683 (eprint:
944	https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2019MS001683) doi:
945	10 1020 /2010MS001683
940	$V_{\rm old}$ din E M (2008) Deletion between temperature consistivity to doubled
947	volodin, E. M. (2008). Relation between temperature sensitivity to doubled
948	L_{resting} $A_{\text{transmiss}}$ and Ω_{restrict} B_{resting} $L_{(2)}$ Detriving 2021 08 04
949	<i>Tzvestiya, Atmospheric and Oceanic Physics, 44</i> (5). Retrieved 2021-08-04,
950	10 1124 /-0001 422000020042 doi:
951	10.1134/\$0001433808030043
952	Voosen, P. (2021, July). U.N. climate panel confronts implausibly hot forecasts of
953	future warming. Retrieved 2021-07-28, from https://www.sciencemag.org/
954	news/2021/07/un-climate-panel-confronts-implausibly-hot-forecasts
955	-future-warming
956	Watanabe, M., & others. (2010). Improved climate simulation by MIROC5: Mean
957	states, variability, and climate sensitivity. J. Climate, 23, 6312–6335.
958	Watanabe, S., Hajima, T., Sudo, K., Nagashima, T., Takemura, T., Okajima, H.,
959	Kawamiya, M. (2011). MIROC-ESM 2010: model description and basic
960	results of CMIP5-20c3m experiments. Geoscientific Model Development, $4(4)$,
961	845–872. doi: 10.5194/gmd-4-845-2011
962	Webb, M., Senior, C., Bony, S., & Morcrette, J. J. (2001). Combining ERBE
963	and ISCCP data to assess clouds in the Hadley Centre, ECMWF and
964	LMD atmospheric climate models. <i>Climate Dyn.</i> , 17, 905–922. doi:
965	10.1007/s003820100157
966	Webb, M. J., Andrews, T., Bodas-Salcedo, A., Bony, S., Bretherton, C. S., Chad-
967	wick, R., Watanabe, M. (2017). The Cloud Feedback Model Intercompar-
968	ison Project (CFMIP) contribution to CMIP6. Geoscientific Model Develop-
969	ment, $10(1)$, $359-384$. doi: $10.5194/\text{gmd}-10-359-2017$
970	Webb, M. J., Lambert, F. H., & Gregory, J. M. (2013). Origins of differences in
971	climate sensitivity, forcing and feedback in climate models. <i>Climate Dynamics</i> ,
972	40(3-4), 677–707. doi: 10.1007/s00382-012-1336-x
973	Williams, I. N., & Pierrehumbert, R. T. (2017). Observational evidence against
974	strongly stabilizing tropical cloud feedbacks. <i>Geophysical Research Let-</i>
975	ters. $\frac{1}{3}$, $\frac{1503-1510}{1503-1510}$. Retrieved 2021-05-04, from https://agupubs
976	.onlinelibrary.wiley.com/doi/abs/10.1002/2016GL072202 (eprint:
977	https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1002/2016GL072202) doi:
079	https://doi.org/10.1002/2016GL072202
070	Williams K. D. Consey, D. Blockley, F. W. Bodas-Salcedo, A
979	Calvert D. Comer B. Xavier P.K. (2018) The Met Of-
980	fice Global Coupled Model 3.0 and 3.1 (GC3.0 and GC3.1) Con-
961	figurations
962	10(2) 357–380 Betrieved 2021-07-17 from https://agupubs
963	onlinelibrary wiley com/doi/abs/10 1002/2017MS001115
904 095	https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1002/2017MS001115) doi:
985	10 1002/2017MS001115 doi: 10.1002/2017MS001115 doi:
980	Wing A A Stauffor C I Booker T Bood K A Abr M S Amold
987	
	N P Zhao M (2020) Clouds and Convective Solf Agreemention
988	N. P., Zhao, M. (2020). Clouds and Convective Self-Aggregation in a Multimodel Ensemble of Badiative Convective Equilibrium Sim
988	N. P., Zhao, M. (2020). Clouds and Convective Self-Aggregation in a Multimodel Ensemble of Radiative-Convective Equilibrium Sim- ulations

991	e2020MS002138. Retrieved 2021-03-25. from https://agupubs
992	.onlinelibrary.wiley.com/doi/abs/10.1029/2020MS002138 (_eprint:
993	https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020MS002138) doi:
994	https://doi.org/10.1029/2020MS002138
995	Young, A. H., Knapp, K. R., Inamdar, A., Hankins, W., & Rossow, W. B. (2018,
996	March). The International Satellite Cloud Climatology Project H-Series
997	climate data record product. Earth System Science Data, $10(1)$, 583–593.
998	Retrieved 2021-03-23, from https://essd.copernicus.org/articles/10/
999	583/2018/ (Publisher: Copernicus GmbH) doi: https://doi.org/10.5194/
1000	essd-10-583-2018
1001	Yukimoto, S., Adachi, Y., Hosaka, M., Sakami, T., Yoshimura, H., Hirabara, M.,
1002	Kitoh, A. (2012). A New Global Climate Model of the Meteorological Re-
1003	search Institute: MRI-CGCM3 — Model Description and Basic Performance—.
1004	Journal of the Meteorological Society of Japan. Ser. II, 90A, 23–64. doi:
1005	10.2151/jmsj.2012-A02
1006	Yukimoto, S., Kawai, H., Koshiro, T., Oshima, N., Yoshida, K., Urakawa, S.,
1007	Ishii, M. (2019). The Meteorological Research Institute Earth System Model
1008	Version 2.0, MRI-ESM2.0: Description and Basic Evaluation of the Physical
1009	Component. Journal of the Meteorological Society of Japan. Ser. II, 97(5),
1010	931–965. doi: $10.2151/jmsj.2019-051$
1011	Zelinka, M. (2021a, August). mzelinka/assessed-cloud-fbks: Aug 16, 2021 Release.
1012	Zenodo. Retrieved 2021-08-16, from https://zenodo.org/record/5206838
1013	doi: 10.5281/zenodo.5206838
1014	Zelinka, M. (2021b, August). mzelinka/cmip56_forcing_feedback_ecs: Aug 16, 2021
1015	Release. Zenodo. Retrieved 2021-08-16, from https://zenodo.org/record/
1016	5206851 doi: 10.5281/zenodo.5206851
1017	Zelinka, M. D., Klein, S. A., & Hartmann, D. L. (2012a). Computing and
1018	Partitioning Cloud Feedbacks Using Cloud Property Histograms. Part I:
1019	Cloud Radiative Kernels. Journal of Climate, $25(11)$, $3715-3735$. doi: 10.1175/:-1: 1.11.00248.1
1020	10.1170/JCII-0-11-00248.1
1021	Zeinika, M. D., Klein, S. A., & Hartmann, D. L. (2012b). Computing and Partition-
1022	Changes in Cloud Amount Altitude and Optical Dopth Lowrad of Climate
1023	25(11) 3736–3754 doi: 10.1175/ICLI-D-11-00249.1
1024	Zelinka M D Klein S A Taylor K E Andrews T Webb M I Cregory
1025	I M & Forster P M (2013) Contributions of Different Cloud Types to
1020	Feedbacks and Bapid Adjustments in CMIP5 Journal of Climate 26(14)
1028	5007-5027. doi: 10.1175/icli-d-12-00555.1
1029	Zelinka, M. D., Myers, T. A., McCoy, D. T., Po-Chedley, S., Caldwell, P. M., Ceppi,
1030	P Taylor, K. E. (2020). Causes of Higher Climate Sensitivity in CMIP6
1031	Models. Geophysical Research Letters, 47(1), e2019GL085782. Retrieved 2020-
1032	12-23, from https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/
1033	2019GL085782 doi: https://doi.org/10.1029/2019GL085782
1034	Zelinka, M. D., Zhou, C., & Klein, S. A. (2016). Insights from a refined decomposi-
1035	tion of cloud feedbacks. Geophysical Research Letters, 43(17), 9259–9269. doi:
1036	10.1002/2016gl069917
1037	Zhu, J., Otto-Bliesner, B. L., Brady, E. C., Poulsen, C. J., Tierney, J. E., Lofver-
1038	strom, M., & DiNezio, P. (2021). Assessment of Equilibrium Climate
1039	Sensitivity of the Community Earth System Model Version 2 Through
1040	Simulation of the Last Glacial Maximum. Geophysical Research Letters,
1041	48(3), e2020GL091220. Retrieved 2021-03-19, from https://agupubs
1042	.onlinelibrary.wiley.com/doi/abs/10.1029/2020GL091220 (_eprint:
1043	https://agupubs.onlinelibrary.wiley.com/doi/pdf/10.1029/2020GL091220) doi:
1044	https://doi.org/10.1029/2020GL091220
1045	Zhu, J., Poulsen, C. J., & Otto-Bliesner, B. L. (2020, May). High climate sensitivity

 1046
 in CMIP6 model not supported by paleoclimate.
 Nature Climate Change,

 1047
 10(5), 378-379.
 Retrieved 2021-01-05, from https://www.nature.com/

 1048
 articles/s41558-020-0764-6
 doi: 10.1038/s41558-020-0764-6

Supporting Information for "Evaluating climate models' cloud feedbacks against expert judgement"

Mark D. Zelinka¹, Stephen A. Klein¹, Yi Qin¹, Timothy A. Myers¹

 $^1 \mathrm{Lawrence}$ Livermore National Laboratory

Contents of this file

1. Figures S1 to S22

Introduction

In this document, we provide 22 supplementary figures. Figures S1 and S2 compare cloud radiative kernel-derived cloud feedbacks with those derived using independent methods. Figure S3 provides a matrix showing which regions and cloud types contribute to each feedback, facilitating understanding of how the assessed feedbacks are computed, which feedbacks are left unassessed, and how we further discretize these remaining unassessed feedbacks. Figures S4-S22 are identical to Figure 1 of the main text, but individual models are highlighted in each.

References

Huang, Y., Xia, Y., & Tan, X. X. (2017). On the pattern of CO2 radiative forcing and poleward energy transport. *Journal of Geophysical Research-Atmospheres*, 122(20), 10578–10593. doi: 10.1002/2017jd027221

- Shell, K. M., Kiehl, J. T., & Shields, C. A. (2008). Using the Radiative Kernel Technique to Calculate Climate Feedbacks in NCAR's Community Atmospheric Model. J. Climate, 21(10), 2269–2282. doi: 10.1175/2007JCLI2044.1
- Soden, B. J., Held, I. M., Colman, R., Shell, K. M., Kiehl, J. T., & Shields, C. A. (2008). Quantifying Climate Feedbacks Using Radiative Kernels. J. Climate, 21, 3504–3520. doi: 10.1175/2007JCLI2110.1
- Taylor, K. E., Crucifix, M., Braconnot, P., Hewitt, C. D., Doutriaux, C., Broccoli, A. J.,
 ... Webb, M. J. (2007). Estimating Shortwave Radiative Forcing and Response in Climate Models. J. Climate, 20(11), 2530–2543. doi: 10.1175/JCLI4143.1
- Zelinka, M. D., Klein, S. A., & Hartmann, D. L. (2012). Computing and Partitioning Cloud Feedbacks Using Cloud Property Histograms. Part I: Cloud Radiative Kernels. *Journal of Climate*, 25(11), 3715–3735. doi: 10.1175/jcli-d-11-00248.1

August 30, 2021, 9:19pm



Figure S1. (left) Zonal and multi-model mean LW and SW cloud feedbacks estimated using three methodologies: cloud radiative kernels (CRK; Zelinka et al., 2012), adjusted change in cloud radiative effect (Δ CRE_{adj}; Soden et al., 2008; Shell et al., 2008), and approximate partial radiative perturbation (APRP; Taylor et al., 2007). Six estimates of Δ CRE_{adj} are shown, each using a different radiative kernel identified in the caption on row 2. (right) Across-model correlation between CRK-derived and non-CRK-derived zonal mean cloud feedbacks. The CRKderived SW cloud feedback is further broken down into optical depth and amount components, which are compared to the APRP-derived SW scattering plus absorption component and amount component, respectively.

August 30, 2021, 9:19pm



Global Mean Cloud Feedbacks

Figure S2. Global mean LW and SW cloud feedbacks estimated using the CRK method scattered against those estimated using non-CRK techniques. For clarity, we show only one of the six estimates of ΔCRE_{adj} , that derived using the kernels of Huang et al. (2017).

Cloud		Amount			Altitude		Optical Depth			
Feedback Components		Ocean		Land	Ocean	Land	Ocean		Land	herese
90°S-90°N	High	N/A		4	1	1	N/A		N/A	Global High ALT Global High ALT Tropical Ocean Descent Low AMT + TAU A Clobal and AMT
	Low	N/A		4	1	1	N/A		N/A	
30°S-30°N	High	Asc 3	Dsc 2	N/A	N/A	N/A	Asc 3	Dsc 2	4	 Global Land AMI Middle latitude Low AMT Extratropical Low TAU
	Low	Asc 3	Dsc 2	N/A	N/A	N/A	Asc 3	Dsc 2	4	lingstorsed
30°-40°N/S	High	8		N/A	N/A	N/A	7		7	 Global Low ALT Tropical Ocean Descent High AMT+TAU Tropical Ocean Ascent Low AMT+TAU Tropical Land High+Low TAU 60-90 Ocean Low AMT 30-40/70-90 Ocean+Land Low TAU 30-90 Ocean+Land High TAU 30-90 Ocean High AMT Global Obscuration Covariance* Global Zelinka et al. (2013) Residual* *not shown in matrix for brevity
	Low	5		N/A	N/A	N/A	6 7		6	
40°-60°N/S	High	8		N/A	N/A	N/A			7	
	Low	5		N/A	N/A	N/A	6		6	
60°-70°N/S	High	8		N/A	N/A	N/A	7		7	
	Low	5		N/A	N/A	N/A	6		6	
70°-90°N/S	High	8	8	N/A	N/A	N/A		7	7	
	Low		5	N/A	N/A	N/A		5	6	N/A = Not Applicable

Figure S3. Matrix of assessed and unassessed cloud feedbacks. The sum of all assessed and unassessed components equals the total cloud feedback.

August 30, 2021, 9:19pm



Figure S4. As in Figure 1, but highlighting CCSM4.





Figure S5. As in Figure 1, but highlighting CanESM2.



Figure S6. As in Figure 1, but highlighting HadGEM2-ES.



X - 9



Figure S7. As in Figure 1, but highlighting MIROC-ESM.



Figure S8. As in Figure 1, but highlighting MIROC5.



Figure S9. As in Figure 1, but highlighting MPI-ESM-LR.



Figure S10. As in Figure 1, but highlighting MRI-CGCM3.



Figure S11. As in Figure 1, but highlighting CNRM-CM6-1.



Figure S12. As in Figure 1, but highlighting CNRM-ESM2-1.



Figure S13. As in Figure 1, but highlighting CanESM5.



Figure S14. As in Figure 1, but highlighting E3SM-1-0.



Figure S15. As in Figure 1, but highlighting GFDL-CM4.



Figure S16. As in Figure 1, but highlighting HadGEM3-GC31-LL.



Figure S17. As in Figure 1, but highlighting IPSL-CM6A-LR.



Figure S18. As in Figure 1, but highlighting IPSL-CM6A-LR-INCA.



Figure S19. As in Figure 1, but highlighting MIROC-ES2L.



Figure S20. As in Figure 1, but highlighting MIROC6.



Figure S21. As in Figure 1, but highlighting MRI-ESM2-0.



Figure S22. As in Figure 1, but highlighting UKESM1-0-LL.